

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Cuba-Borda, Pablo; Singh, Sanjay R.

Working Paper Understanding persistent ZLB: Theory and assessment

Working Paper, No. 346

Provided in Cooperation with: University of California Davis, Department of Economics

Suggested Citation: Cuba-Borda, Pablo; Singh, Sanjay R. (2021) : Understanding persistent ZLB: Theory and assessment, Working Paper, No. 346, University of California, Department of Economics, Davis, CA

This Version is available at: https://hdl.handle.net/10419/267010

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

Understanding Persistent ZLB: Theory and Assessment^{*}

Pablo Cuba-Borda

Federal Reserve Board

Sanjay R. Singh University of California, Davis

December 2021

Abstract

Concerns of prolonged stagnation periods with near-zero interest rates and deflation have become widespread in many advanced economies. We build a theoretical framework that rationalizes two theories of low interest rates: *expectations-trap* and *secular stagnation* in a unified setting. We analytically derive contrasting policy implications under each hypothesis and identify robust policies that eliminate expectations-trap and reduce the severity of secular stagnation episodes. We provide a quantitative assessment of the Japanese experience from 1998:Q1-2020:Q4. We find evidence favoring the expectations-trap hypothesis and show that equilibrium indeterminacy is essential to distinguish between theories of low interest rates in the data.

Keywords: Expectations-driven trap, secular stagnation, zero lower bound, robust

policies.

JEL Classification: E31, E32, E52.

^{*}Correspondence: P Cuba-Borda: Division of International Finance, Federal Reserve Board, 20th Street & Constitution Ave., NW, Washington, DC 20551. Email: pablo.a.cubaborda@frb.gov. SR Singh: Department of Economics, University of California, Davis, CA 95616. Email: sjrsingh@ucdavis.edu. This paper replaces a previous version circulated with the title "Understanding Persistent Stagnation". We thank Borağan Aruoba, Florin Bilbiie, Carlos Carvalho, Gauti Eggertsson, Pascal Michaillat, Jean-Baptiste Michau, Taisuke Nakata, Giovanni Nicolò, Sebastian Schmidt, Kevin Sheedy and participants at various presentations for helpful comments and suggestions. Disclaimer: The views expressed in this article are solely the responsibility of the authors and should not be interpreted as reflecting the views of the Board of Governors of the Federal Reserve System or of anyone else associated with the Federal Reserve System.

"I believe that for the euro area there is some risk of *Japanification*, but it is by no means a foregone conclusion." — Mario Draghi (January, 2020).

1. INTRODUCTION

Since the global financial crisis of 2008-2009, concerns of prolonged near-zero interest rates and meager inflation became predominant across many advanced economies, most notably in Europe and the United States. Such concerns, dubbed as *Japanification*, relate to the decades-long stagnation of the Asian economy following the collapse of a real-estate bubble in the early 1990s. As a result, nominal interest declined to zero, and deflation emerged, leaving the central bank unable to fight recessions.¹

Two predominant hypotheses rationalize interest rates near zero and inflation below the central bank's target. The first hypothesis is that of *expectation-driven* liquidity trap whereby pessimistic deflationary expectations become self-fulfilling in the presence of the zero lower bound (ZLB) constraint on short-term nominal interest rates (Benhabib, Schmitt-Grohé and Uribe, 2001, 2002). The second hypothesis is that of *secular stagnation* that entails a persistently negative natural interest rate constraining the central bank at the ZLB (Hansen, 1939; Summers, 2013). The theory and policy implications of these hypotheses have been developed in the literature using different frameworks.² Our unifying framework bridges this gap, facilitating analytical comparison and quantitative assessments.

In this paper, we build a theoretical framework that rationalizes *expectation-driven* liquidity traps and *secular stagnation* in a unified setting. We analytically derive contrasting policy implications under each hypothesis. For example, a payroll tax cut exacerbates the recession under secular stagnation but is expansionary in the expectations-trap model. Since conventional policy measures may have opposite

¹Financial Times, "Japanification: investors fear malaise is spreading globally," August 26, 2019. ASSA Annual Meeting Panel Session: "Japanification, Secular Stagnation, and Fiscal and Monetary Policy Challenges," January 2020.

²Expectations traps have been investigated using representative agent models (Benhabib et al., 2001; Schmitt-Grohé and Uribe, 2017), while secular stagnation is typically modeled using the overlapping generation framework (Eggertsson and Mehrotra, 2015; Eggertsson, Mehrotra, Singh and Summers, 2016).

implications, there is a need for robust policies to deal with liquidity traps (Bilbiie, 2018; Nakata and Schmidt, 2021). Our theoretical framework allows us to identify a set of robust policies that operate by placing a sufficiently high floor on the inflation rate. Imposing a lower bound on the inflation rate excludes the possibility of an expectations-driven liquidity trap. Policies that put a bound on deflation also reduce the severity of liquidity traps induced by exogenous declines in the natural rate of interest.

To obtain analytical results, we modify the textbook New Keynesian with endogenous discounting (Uzawa, 1968; Epstein, 1983). As a result, the representative agent's Euler equation in a steady-state equilibrium features a negative relationship between output and real interest rate, similar to a static IS curve. This modification breaks the tight connection between the natural interest rate and the discount factor, thus allowing for a permanently negative natural interest rate. On the supply side, we introduce tractable nominal frictions to obtain a linear relationship between inflation and output (Bhattarai, Eggertsson and Gafarov, 2019). Within this framework, we prove the existence of three steady-state equilibria. A *targeted-inflation* steady state at which the central bank can meet its inflation target, the economy is at full employment, and the nominal interest rate is positive. In addition, there are two liquidity trap steady states at which inflation is below the central bank's intended target. The level of output is below full employment, and the nominal interest rate is at the ZLB. Both liquidity trap steady states are the focus of our analysis.

The liquidity traps steady states arise in the presence of the ZLB constraint on shortterm nominal interest rates. In one case, combined with long-run money non-neutrality, a shift in agent's inflation expectations makes the liquidity trap equilibrium selffulfilling. For this reason we label it as *expectations-driven* liquidity trap. Alternatively, when prices are rigid, the economy can settle in a liquidity trap because of a permanent decline in the natural interest rate. In this case, the liquidity trap arises because of a change in the economy's fundamentals and not a shift in expectations. For this reason we label this situation as a *secular stagnation* steady-state or *fundamentals-driven* liquidity trap. In the absence of discounting, the natural rate of interest is constant and equal to the inverse of the household's discount factor, and the model cannot accommodate the *secular stagnation* hypothesis.

We use the Japanese experience from 1998-2020 as a laboratory to contrast the two hypotheses and offer the first quantitative assessment of expectation-driven liquidity traps versus secular stagnation. We embed our theory in a quantitative New Keynesian model and assess if a policymaker can use the data to discern the predominant hypothesis. Using bayesian prediction pools, we estimate the probabilistic assessment of the relevant model in real-time (Geweke and Amissano, 2011; Del Negro, Hasengawa and Schorfheide, 2016). Our quantitative analysis offers two main findings. First, we find evidence that Japan is more likely to be in an expectations-driven liquidity trap. Second, there is considerable real-time uncertainty in discerning between secular stagnation and the expectations-driven trap models, especially during Japan's first decade of near-zero interest rates.

We find that equilibrium indeterminacy is central to tilt our quantitative assessment in favor of the expectations-trap hypothesis. This result emerges because the dynamic properties of the ZLB equilibrium differ across the two narratives. Under secular stagnation, the ZLB equilibrium exhibits locally determinate dynamics. In contrast, the expectation traps model features locally indeterminate dynamics around the ZLB steady state. Thus the equilibrium dynamics are consistent with a multiplicity of stable paths. Because our quantitative analysis focuses on a long-lasting ZLB episode, equilibrium selection implies restrictions for the response of output and inflation to structural disturbances. Using procedures that maximize the model likelihood, we let the data select the best-fitting equilibrium. At the same time, our Bayesian procedure intrinsically penalizes the likelihood function for the presence of additional parameters in the expectations-trap equilibria (Schwarz, 1978; Lubik and Schorfheide, 2004).

What accounts for the better fit of the expectations-trap hypothesis in Japan? We find that the negative correlation between output growth and inflation in Japanese data is a central empirical moment for equilibrium selection and model fit. We find that the equilibrium dynamics around the secular stagnation steady state cannot deliver the observed negative correlation. With interest rates pegged at the ZLB, any shock that

generates a persistent increase in inflation rate lowers the real interest rate, increases consumption, and therefore output and inflation positively co-move. In contrast, local indeterminacy of the expectations-trap steady state implies that inflation can adjust in any direction. Our estimation procedure allows the data to pin down this response. The result is that expectation traps can generate an unconditional correlation of inflation and output close to that observed in the data.

We further investigate our empirical results along three dimensions. First, we restrict equilibrium selection using the minimal state variable (MSV) criterion (McCallum, 2004). We analytically show that the MSV solution implies a positive co-movement between inflation and output under expectations-trap. In this case, the prediction pool analysis cannot distinguish between secular stagnation and expectations-driven liquidity traps from the data. Second, we investigate the importance of non-fundamental i.i.d. shocks—known as sunspots—that emerge due to indeterminate model dynamics in the expectations-trap model. Our benchmark results index the multiplicity of equilibrium through the correlation between fundamental and sunspot shocks using the method of Bianchi and Nicolo (2021). We find that restricting the correlation between price-markup and sunspot shocks to non-negative values worsens the fit of the expectations-trap model and favors secular stagnation. Thus, using data to discipline equilibrium selection is central to our results. Third, we verify that the expectations-trap hypothesis generates a negative correlation between inflation and output in a calibrated medium-scale new Keynesian model, while the secular stagnation hypothesis does not. This final exercise implies that our analytical insights carry over to a wide class of models commonly used for policy analysis.

Relation to the literature. Our work complements the recent analyses of Michaillat and Saez (2021), Michau (2018), and Ono and Yamada (2018) who use the bonds-inutility assumption to analyze a unique secular stagnation scenario.³ We distinctly focus on understanding the differences between the two stagnation concepts analytically and quantitatively. These alternate micro-foundations essentially introduce discounting

³Following Feenstra (1986) and Fisher (2015), a functional equivalence can be shown between using bonds in the utility and endogenous discounting.

into the Euler equation. Our paper jointly considers the two narratives of persistent ZLB and offers quantitative and analytical insights.

This paper is also related to the work by Mertens and Ravn (2014), Aruoba, Cuba-Borda and Schorfheide (2018), Bilbiie (2018), and Nakata and Schmidt (2021), who contrast expectations-driven and fundamental-driven liquidity traps using the standard Euler equation without discounting. Their setup can only accommodate a short-lived fundamentals-driven liquidity trap, while our modified Euler equation allows the possibility of secular stagnation as a competing hypothesis. Our paper is also complementary to Schmitt-Grohé and Uribe (2017), which analyzes the case of permanent expectations-driven liquidity traps. Coyle and Nakata (2019) characterizes optimal inflation target in the presence of expectations-liquidity traps. We build on these papers to show that policies that impose a lower bound on inflation preclude the expectations-driven traps. Benigno and Fornaro (2018)'s stagnation-trap, which focuses on the role of pessimism about the economy's growth rate, is complementary to the inflation pessimism we evaluate in this paper.⁴

Our framework allows agents in the model to expect ZLB episodes of permanent duration under both hypotheses. This feature stands in contrast to models that use transitory declines in the natural interest rate to generate ZLB episodes where agents' expectations have to be consistent with recovery to the full-employment steady state in the medium run (Bianchi and Melosi, 2017; Nakata, 2017; Nakata and Schmidt, 2019). Moreover, when modeling temporary liquidity traps, there is an equilibrium selection rule imposed, implicitly, by the assumed behavior of inflation at the end of the liquidity trap (Cochrane, 2017). We sidestep this issue by considering permanent ZLB episodes. Under secular stagnation, equilibrium dynamics are locally determinate. For expectations-driven liquidity traps, we use data to discipline equilibrium selection explicitly through the model's likelihood.

Our static prediction pool analysis is related to Lansing (2019) in which a model with endogenous switching regimes generates data from a time-varying mixture of

⁴As they highlight, the possibility of a self-fulfilling expectations trap is more likely when multiple sources of pessimism (growth, deflation) are allowed in the same model. We leave the analysis of such models to future research.

two models. In this paper, we construct a time-varying probability on the predictive densities coming from two alternative models. Our paper also relates to Mertens and Williams (2021) that use the implications of changes in the natural interest rate on the distribution of interest rates and inflation in the options data from the U.S. financial markets, to discern between fundamentals- and expectations-driven liquidity. Instead, we explore the implications of changes in government spending, technology growth, and markups. We use the principal-agent decision framework of Del Negro et al. (2016) to identify the relevant hypothesis in Japan, combining the predictive densities derived from consumption, output and inflation data. The Bayesian nature of our approach allows us to measure the uncertainty about the contrasting hypotheses with a structural model.

A common theme of the papers that study expectations-driven liquidity traps is that policy implications may be opposite of the ones derived from fundamentals-driven liquidity traps. It becomes central, as a result, to assess what hypothesis is dominant in the data and come up with policies that can always be stabilizing.⁵ Our robust policies prescribe a flattening of the aggregate supply curve to preclude the existence of expectations-driven liquidity traps. Our analysis is related to fiscal policy rules that prevent the decline of real marginal costs (Schmidt, 2016) or fiscal stabilization policies that eliminate expectation-traps (Nakata and Schmidt, 2021).⁶ Similarly, research and development (R&D) subsidies advocated by Benigno and Fornaro (2018) that affect aggregate supply in an endogenous growth environment can eliminate expectations-driven liquidity traps.

Layout. Section 2 presents the main theoretical results of the paper. Section 3 presents the quantitative model to assess the Japanese experience, and Section 4 presents our quantitative findings. In Section 5 we investigate the role of equilibrium selection. Section 6 extends our analysis to a calibrated medium-scale DSGE environment. Section 7 concludes. All proofs and additional results are in the online appendix.

⁵One can develop expectations-traps equilibria with similar comparative statics as the fundamentalsdriven liquidity traps. Our analysis does not focus on those, as they may not generate policy dilemmas.

⁶Our minimum wage policy is also related to the work of Glover (2019) which introduces a tradeoff for employment stability through an allocative inefficiency.

2. Key insights in a two equations setup

We begin with a simple setup that analytically demonstrates our model's ability to entertain the expectations-driven liquidity trap and the fundamentals-driven liquidity trap. There are two main takeaways: a) a high degree of nominal rigidity may prohibit the existence of expectation traps, and b) a fundamentals-driven liquidity trap can exist for an arbitrarily long duration. The model features preferences with endogenous discounting and a particular variant of price setting by monopolistically competitive firms that generates analytical results. We characterize and formally define the different steady states: *targeted inflation* state, *secular stagnation* steady state, and the *expectations-driven trap* (also referred to as the BSGU steady state).

2.1. Household

Time is discrete and there is no uncertainty. For now, there is no government spending. A representative agent maximizes the following:

$$\max_{\{C_t, b_t\}} \sum_{t=0}^{\infty} \Theta_t \left[\log C_t - \omega h_t \right]$$
$$\Theta_0 = 1; \quad \Theta_{t+1} = \hat{\beta}(\tilde{C}_t) \Theta_t \ \forall t \ge 0$$

where Θ_t is an endogenous discount factor (Uzawa, 1968; Epstein and Hynes, 1983), C_t is consumption, \tilde{C}_t is average consumption that the household takes as given, and h_t is hours. Such preferences are prominent in the small open economy literature (Schmitt-Grohé and Uribe, 2003).⁷

For tractability, we assume a linear functional form for $\hat{\beta}(\cdot) = \delta_t \beta \tilde{C}_t$, where $0 < \beta < 1$ is a parameter, \tilde{C}_t is average consumption that the household takes as given, and $\delta_t > 0$ are exogenous shocks to the discount factor. In contrast to the conventional assumption in the endogenous discounting, we require $\hat{\beta}' > 0$. This is often referred to

⁷What will be essential for our purpose is that there is a negative steady state relationship between consumption and real interest rate to generate discounting in the Euler equation. We can show a functional equivalence between preferences with endogenous discounting and a recent approach that employs bonds in utility (Michaillat and Saez, 2021).

as decreasing marginal impatience (DMI) in the literature.⁸

The household earns wage income $W_t h_t$, interest income on past bond holdings of risk-free government bonds b_{t-1} at gross nominal interest rate R_{t-1} , dividends Φ_t from firms' ownership and makes transfers T_t to the government. Π_t denotes gross inflation rate. The period by period (real) budget constraint faced by the household is given by $C_t + b_t = \frac{W_t}{P_t}h_t + \frac{R_{t-1}}{\Pi_t}b_{t-1} + \Phi_t + T_t$. An interior solution to household optimization yields the consumption Euler equation, and intra-temporal labor supply condition

$$1 = \beta(\tilde{C}_t) \left[\frac{C_t}{C_{t+1}} \frac{R_t}{\Pi_{t+1}} \right]; \quad \frac{W_t}{P_t} = \omega C_t$$

In equilibrium, individual and average consumption are identical, i.e. $C_t = \tilde{C}_t$. The Euler equation simplifies to:

$$1 = \delta_t \beta C_t \left[\frac{C_t}{C_{t+1}} \frac{R_t}{\Pi_{t+1}} \right]$$

Discussion of DMI:

The DMI assumption implies that as agents get wealthier, they become more patient. Several papers in the literature agree it is more realistic and intuitive to assume decreasing impatience rather than increasing impatience (Epstein and Hynes, 1983; Uzawa, 1968). Despite the realism, it is conventional to use increasing marginal impatience since it is consistent with the boundedness of wealth and stability in environments with an exogenous real interest rate. As shown by Das (2003), if the returns to savings diminish at a high enough rate, it is possible to guarantee stability in environments with decreasing marginal impatience.⁹

⁸Our assumed functional form can be considered a special case of the more general functional: $\hat{\beta}(\cdot) = \delta_t \beta C_t^{\gamma_c}$. When $\gamma_c = 0$, this nests the textbook case of exogenous discounting. We consider $\gamma_c = 1$ for tractability.

⁹Following the insights of Das (2003), the decreasing marginal impatience assumption is consistent with the existence of stable equilibrium dynamics with capital accumulation. Results are available upon request.

2.2. Production

A perfectly competitive final-good producing firm combines a continuum of intermediate goods indexed by $j \in [0,1]$ using the CES Dixit-Stiglitz technology: $Y_t = \left(\int_0^1 Y_t(j)^{1-\nu} dj\right)^{\frac{1}{1-\nu}}$, where $1/\nu > 1$ is the elasticity of substitution across varieties. The price of the final good $P_t = \left(\int_0^1 P_t(j)^{\frac{\nu-1}{\nu}} dj\right)^{\frac{\nu}{\nu-1}}$. Profit maximization gives The demand for intermediate good j can be derived from profit maximization as $Y_t(j) = \left(\frac{P_t(j)}{P_t}\right)^{-1/\nu} Y_t$.

Intermediate good *j* is produced by a monopolist with a linear production technology: $Y_t(j) = h_t(j)$. Intermediate good producers buy labor services $H_t(j)$ at a nominal price of W_t . Moreover, they face nominal rigidities in terms of price adjustment costs, and maximize profits $\Phi_t(j) = (1 + \tau)P_t(j)Y_t(j) - W_th_t(j)$, where τ is a production subsidy.

We introduce nominal rigidities in the price-setting decision of these intermediate producers. To derive a tractable Phillips curve, we follow Bhattarai et al. (2019). A fraction α_p of the firms set prices flexibly every period to maximize per period profits $(1 + \tau)\frac{p_t^*}{P_t} = \frac{1}{1-\nu}\frac{W_t}{P_t}$. We set the production subsidy to eliminate markups so that (imposing $Y_t = C_t$), optimal price becomes $\frac{p_t^*}{P_t} = \omega Y_t$. The remaining fraction $1 - \alpha_p$ index their price p^n to the aggregate price level from previous period $\frac{p_t^n}{P_t} = \Gamma_t \frac{P_{t-1}}{P_t}$ where Γ_t is indexing variable to be defined shortly. The price index then becomes $P_t^{\frac{\nu-1}{\nu}} = \alpha_p (p_t^*)^{\frac{\nu-1}{\nu}} + (1 - \alpha_p) (p_t^n)^{\frac{\nu-1}{\nu}}$. With choice of $\nu = 1/2$ and $\Gamma_t = \frac{P_t}{Y_t^{-1}(P_t - P_{t-1}) + P_{t-1}}$, we can derive the following relationship between gross inflation and aggregate output:¹⁰

$$\Pi_t = \kappa Y_t + (1 - \kappa \bar{Y})$$

where $\kappa \equiv \frac{\alpha_p}{1-\alpha_p}$ is slope of the Phillips curve and \bar{Y} is steady state output in the absence of nominal rigidities (or zero price dispersion). We set $\omega = 1$ so as to normalize $\bar{Y} = 1$.

¹⁰With $\Gamma_t = 1$, we get the neoclassical Phillips curve (See Ch 3.1 Woodford (2003)). Allowing for indexation to depend on current output allows us to derive Phillips curve that helps us derive insights analytically and make comparison with downward nominal wage rigidity assumption. More generally, the firms that are indexing prices follow an indexation rule $\Gamma(Y_t)$ where $\Gamma'(\cdot) > 0$ i.e. price reduction is increasing in unemployment. Similar analytical results with $\Gamma_t = Y_t$ can be shown.

2.3. Government and resource constraint

We close the model by assuming a government that balances budget every period and a monetary authority that sets the nominal interest rate on nominal risk-free one-period bonds using the following Taylor rule $R_t = \max\{1, (1 + r^*)\Pi_t^{\phi_{\pi}}\}$, where $1 + r_t^* \equiv \frac{1}{\delta_t\beta}$ is the natural interest rate, and $\phi_{\pi} > 1$. In equilibrium, bonds are in zero net supply. ¹¹ We implicitly assumed the (gross) inflation target of the central bank to be one. The zero lower bound (ZLB) constraint on the short-term nominal interest rate introduces an additional non-linearity in the policy rule. Finally, we assume that the resource constraints hold in the aggregate: $C_t = Y_t$, and $b_t = 0$.

2.4. Equilibrium

The competitive equilibrium is given by the sequence of three endogenous processes $\{Y_t, R_t, \Pi_t\}$ that satisfy the conditions 1–3 for a given exogenous sequence of process $\{\delta_t\}_{t=0}^{\infty}$ and initial price level P_{-1} :

$$1 = \delta_t \beta Y_t \left[\frac{Y_t}{Y_{t+1}} \frac{R_t}{\Pi_{t+1}} \right]$$
(1)

$$\Pi_t = \kappa Y_t + (1 - \kappa) \tag{2}$$

$$R_t = \max\{1, (1 + r_t^*)\Pi_t^{\phi_{\pi}}\}$$
(3)

where the exogenous sequence of natural interest rate is given by $1 + r_t^* \equiv \frac{1}{\delta_t \beta}$.

2.5. Non-stochastic steady state

We can represent the steady-state equilibrium with an aggregate demand block and an aggregate supply block.

Aggregate Demand (AD) is a relation between output and inflation and is derived by

¹¹The natural interest rate is the real interest rate on one-period government bonds that would prevail in the absence of nominal rigidities.

combining the Euler equation and the Taylor rule. The AD curve is given by

$$Y_{AD} = \frac{1}{\beta \delta} \begin{cases} \frac{1}{(1+r^*)\Pi^{\phi_{\pi}-1}}, & \text{if } R > 1, \\ \Pi, & \text{if } R = 1 \end{cases}$$
(4)

When the ZLB is not binding, the AD curve has a strictly negative slope; and it becomes linear and upward sloping when the ZLB constrains the nominal interest rate. Thus, the kink in the aggregate demand curve occurs at the inflation rate where the ZLB constrains monetary policy: $\Pi_{kink} = \left(\frac{1}{(1+r^*)}\right)^{\frac{1}{q_{\pi}}}$. When $1 + r^* > 1$, the kink in the AD curve occurs at an inflation rate below 1. For the natural interest rate to be positive, the patience parameter must be low enough i.e. $\delta < \frac{1}{\beta}$.

Aggregate Supply (AS) is given by $\Pi = \kappa Y + (1 - \kappa)$ in the steady state. When Y = 1, $\Pi = 1$. In this linear aggregate supply curve, the degree of nominal rigidity κ also determines the lower bound on inflation = $1 - \kappa$. In the quantitative section, we will work with the standard forward-looking NK Phillips curve.

In this two-equation representation, we can characterize and prove the existence of different steady-state equilibria. Proposition 1 shows that a targeted steady state exists as long as the natural interest rate is positive.

Proposition 1. (Targeted Steady State): Let $0 < \delta < \frac{1}{\beta}$. There exists a unique positive interest rate steady state with Y = 1, $\Pi = 1$ and $R = \frac{1}{\beta\delta} > 1$. It features output at efficient steady state, and inflation at the policy target. The equilibrium dynamics in this steady state's neighborhood are locally determinate.

A steady state at which the central bank can meet its inflation target is *defined* as the *targeted-inflation steady state*. The presence of a targeted-inflation steady state is contingent on the natural interest rate and the monetary authority's inflation target. With a unitary inflation target, it must be the case that the natural interest rate is non-negative, which is implied by the assumption of $\delta < \frac{1}{\beta}$. In Proposition 2 we show that, a liquidity trap steady state (à la Schmitt-Grohé and Uribe, 2017) may jointly co-exist with the targeted steady state described above. However, with a flat enough





Phillips curve, a targeted steady state is the unique steady state in this economy. A high enough nominal rigidity prevents inflation from falling to levels such that self-fulfilling deflationary expectations do not manifest in the steady state.

Proposition 2. (Expectations trap steady state): Let $0 < \delta < \frac{1}{\beta}$. For $\kappa > 1$ (i.e. $\alpha_p > 0.5$) there exist two steady states:

- 1. The targeted steady state with Y = 1, $\Pi = 1$ and $R = \frac{1}{\beta\delta} > 1$.
- 2. (*Expectations-driven* trap) A unique-ZLB steady state with $Y = \frac{1-\kappa}{\beta\delta-\kappa} < 1$, $\Pi = \frac{\beta\delta(1-\kappa)}{\beta\delta-\kappa} < 1$ and R = 1. The local dynamics in a neighborhood around this steady state are locally indeterminate.

When prices are rigid enough, i.e., $\kappa < 1$, there exists a unique steady state, and it is the targeted inflation steady state. When prices are flexible $\alpha_p = 1$ ($\kappa \rightarrow \infty$), two steady states exist. A unique deflationary steady state with zero nominal interest rates and a unique targeted inflation steady state.

Panel a) in Figure 1 illustrates the unique targeted-steady state (Y^f, Π^f) and the unique ZLB steady state (Y^b, Π^b) with the modified Euler equation. We define the *expectations-driven* trap as the steady state with a positive natural interest rate, negative output gap, and deflation and in whose neighborhood the equilibrium dynamics are

locally indeterminate. Pessimistic inflationary expectations can push the economy to this steady state without any change in fundamentals.

We now consider the case where adverse fundamentals can push the economy to a permanent liquidity trap. If agents are sufficiently patient $\delta > \frac{1}{\beta}$, i.e., the natural rate of interest is negative, and the ZLB constrains monetary policy. In that case, the nominal interest rate is permanently zero while there is below-potential output and deflation in the economy. We characterize this possibility in Proposition 3.¹²

Proposition 3. (Secular Stagnation): Let $\delta > \frac{1}{\beta}$ and $\kappa < 1$. There exists a unique steady state with $Y = \frac{1-\kappa}{\beta\delta-\kappa} < 1$, $\Pi = \frac{\beta\delta(1-\kappa)}{\beta\delta-\kappa} < 1$ and R = 1. It features output below the targeted steady state and deflation, caused by a permanently negative natural interest rate. The equilibrium dynamics in this steady state's neighborhood are locally determinate.

See panel b) in Figure 1 for illustration of this unique steady state. The intersection of the solid red line(AD) with the solid blue line (AS) at (Y^f, Π^f) depicts the result of proposition 1, and the intersection of dashed red and blue lines at (Y^s, Π^s) depicts the liquidity trap steady state in proposition 3. We formally *define* the *secular stagnation steady state* as the steady state featuring negative output gap, zero nominal interest rate on short-term government bonds and exhibiting *locally determinate* equilibrium dynamics in its neighborhood. This local determinacy property is the main difference between the secular stagnation narrative and the expectations-driven narrative.

Note that the secular stagnation steady state exists in this model because of sufficient discounting in the modified Euler equation. Unlike the traditional new Keynesian model, an arbitrarily long ZLB episode driven by a negative natural rate can exist in equilibrium. In log-linearized new Keynesian models without discounting, *deflationary black holes* emerge as the duration of the temporary liquidity trap is increased, with inflation and output tending to negative infinity (Eggertsson, 2011). The solution remains bounded in our setup as the duration of ZLB is increased.

¹²Note the efficient steady state is always an equilibrium of an economy without nominal rigidities.

2.6. Comparative Statics

The expectations-trap steady state and the secular stagnation steady state have different implications for policy. At the ZLB, a leftward shift in the aggregate demand graph lowers output and inflation under secular stagnation, but it is expansionary under expectations-trap. Similar policy reversals emerge due to a permanent increase in the nominal interest rate and positive supply shocks. We discuss comparative statics on the rise in the nominal interest rate and labor tax cuts. Because of the local determinacy properties of the secular stagnation steady state, the comparative static experiment is well-defined without the need for additional assumptions. In contrast, for an expectations trap, we assume that inflation expectations do not change drastically to push the economy to the full-employment steady state in response to the experiment.

2.6.1 Neo-Fisherian Exit

We now discuss the effects of a permanent increase in nominal interest rate. We model the policy as a permanent change in the intercept of the Taylor rule, *a*: $R^{new} = \max\{1 + a, a + R^* \left(\frac{\Pi}{\Pi^*}\right)^{\alpha_{\pi}}\} = a + R$, where *a* is increased to a positive number from zero, this policy simultaneously increases the lower bound on the nominal interest rate. It thus does not have any effect on the placement of the kink in the aggregate demand curve. it acts as a shifter for aggregate demand graph which is now given by: $Y_{AD} = \frac{1}{\beta\delta} \frac{\Pi}{a+R}$.

Given the inflation rate, an increase in *a* lowers the quantity of output demanded. This change induces deflationary pressures at the secular stagnation steady state. Lower inflation then increases the real interest rate gap and causes a further drop in production. In contrast, during an expectations trap, an increase in nominal interest rate anchors agents' expectations to higher levels of inflation, thus obtaining neo-Fisherian results (Schmitt-Grohé and Uribe, 2017; Uribe, 2021; Schmitt-Grohé and Uribe, Forthcoming).

2.6.2 Labor tax cuts

We now show the effects of a permanent reduction in payroll taxes financed by increase in lump-sum taxes. This policy reform is an aggregate supply shifter. Let workers' take home (real) wages be $(1 - \tau^w) \frac{W_t}{P_t}$ where τ^w are payroll taxes. In the steady state, aggregate supply curve is now given by: $\Pi = \kappa Y_{AS} + (1 - \frac{\kappa}{1 - \tau^w})$.

Given the inflation rate, a reduction in τ^w increases workers' labor supply. The tax reduction induces deflationary pressures in the secular stagnation steady state, lower inflation increases the real interest rate gap and causes a further drop in output. This result corresponds to Eggertsson (2010)'s *paradox of toil*. In contrast, during an expectation trap, the increase in output dominates the deflationary pressures since the aggregate supply graph is steeper than the aggregate demand graph. Reductions in payroll taxes are expansionary in the case of an expectation trap. Similar results apply for structural reforms that reduce intermediate goods' markups (Eggertsson, Ferrero and Raffo, 2014).

2.7. A robust policy: appropriate price indexation

The disparate policy implications across the two steady states motivate the need for developing policies that may be robust to the source of recession. We introduce one such set of policy prescriptions to tackle these recessions. When a liquidity trap is transitory, inflationary pressures from a rise in price markups can be expansionary (Eggertsson, 2012). We use this insight to show that appropriate price indexation rule can increase output under secular stagnation and eliminate the expectation traps.

We prove that an appropriate indexation scheme can eliminate expectations-driven liquidity trap while also improving the output under secular stagnation. Recount that a fraction $1 - \alpha_p$ of firm index their price p^n to the aggregate price level from previous period $\frac{p_t^n}{P_t} = \Gamma_t \frac{P_{t-1}}{P_t}$. We now allow Γ_t to be somewhat general: $\Gamma_t = \frac{P_t}{Y_t^{-1}(P_t - \lambda P_{t-1}) + P_{t-1}}$; $\lambda > 0$. The price Phillips curve is given by: $\Pi_t = \kappa Y_t + (\lambda - \kappa \bar{Y})$. We consider λ to be a policy tool that requires adjusting firms to index prices in a particular manner. **Proposition 4.** Consider an indexation rule where the non-resetters index their prices to last period's price level with indexation coefficient: $\Gamma_t = \frac{P_t}{Y_t^{-1}(P_t - \lambda P_{t-1}) + P_{t-1}}$. There does not exist expectations-driven liquidity trap $\forall \lambda > \kappa$. Output and inflation under secular stagnation are increasing in λ .

Under secular stagnation, higher values of λ act like higher price-markups, and increase output. A policy setting $\lambda > \kappa$ delivers a lower bound on deflation eliminating expectations-driven liquidity traps. Other policies that flatten the Phillips curve by strengthening labor unions during recessions can also preclude the possibility of expectation traps as well. A converse implication of this finding is that structural reforms that increase downward flexibility in prices make the economy vulnerable to swings in pessimistic expectations. We label this result as the *curse of flexibility*.¹³

2.8. Robust policy with downward nominal wage rigidity

We briefly discuss the robustness of our theoretical results with downward nominal wage rigidity. In particular, we discuss that minimum wage policies can also act like robust policies.¹⁴ We make two changes to the model presented in Section 2. One, we assume an inelastic labor supply with a time endowment of one. Second, we assume nominal wages are downwardly rigid as in Schmitt-Grohé and Uribe (2017) i.e. $W_t \ge \gamma(h_t)W_{t-1}$, where $\gamma(0) = 1 - \kappa$ is a constant, $\gamma'(\cdot) > 0$, and $\gamma(1) > \delta\beta$. By choosing $\kappa < 1$, a policymaker can eliminate the expectations-driven liquidity trap. Since this policy lever is about the level of wages paid when employment approaches zero (an off-equilibrium limit), our model implies that policies similar to a *universal basic income*, can help fight expectations-driven recessions. We provide proof for these statements in Appendix F.

¹³Our analysis does not imply that imposing a lower bound on deflation is enough to eliminate the expectations-trap steady state in more general settings. For example, in Benigno and Fornaro (2018) there is perfect downward nominal rigidity, but endogenous growth opens up the possibility of a stagnation trap. In Heathcote and Perri (2018), despite the presence of perfectly downward rigid wages, an expectation-trap steady state exists because of the precautionary savings motive.

¹⁴Furthermore, in Appendix G, we show that the expectations-driven trap can also emerge in the overlapping generations model of Eggertsson, Mehrotra and Robbins (2019) with appropriate wage flexibility.

3. A QUANTITATIVE EXPLORATION

We now present a quantitative analysis based on a small-scale New Keynesian model that has been widely studied in the literature—see An and Schorfheide (2007). The critical difference is an Euler equation that features discounting. Relative to Section 2, we introduce a forward-looking Phillips curve and exogenous shocks to government spending, technology growth, and price-markups. Because the model is relatively standard, we focus on the log-linearized equilibrium conditions. We discuss the calibration that gives rise to the two ZLB hypotheses and discuss our strategy to take the model to the data.

3.1. Modified Euler Equation

The central piece that generates secular stagnation steady state is the modified Euler equation. As shown in equation 5, we use an specification that features an additive wedge, that arises from a bonds-in-utility specification, instead of the multiplicative wedge considered in Section 2.

$$1 = \beta \left(\frac{c_{t+1}}{c_t}\right)^{-1} \frac{R_t}{z_{t+1}\Pi_{t+1}} + \delta c_t \tag{5}$$

The term $\delta \ge 0$ corresponds to marginal utility of holding bonds (see Appendix C for derivation). We have three reasons for such choice: a) as $\delta \to 0$, this equation nests the textbook Euler equation as a special case; b) similar wedge can be derived from a wealth-in-utility argument (Michaillat and Saez, 2021), which is an alternate device to generate a persistent fundamentals-driven liquidity trap; and c) the additive wedge is related to models that emphasize flight-to-liquidity aspects of the Great Recession (Del Negro, Eggertsson, Ferrero and Kiyotaki, 2017).¹⁵ We use the parameter δ to target empirical estimates of the natural interest rate in Japan. The calibration will depend on the particular hypothesis, and we describe our strategy shortly later in this section.

¹⁵Some of the analytical results in Section 2 can be shown with this additive wedge. We discuss a comparison with the standard Euler equation in Appendix H.

3.2. Equilibrium Conditions

We approximate the equilibrium conditions around a permanent liquidity trap steady state. Let \hat{y}_t , $\hat{\pi}_t$ and \hat{c}_t denote the log-deviations of output, inflation and consumption, respectively, relative to the steady state of interest. The following equations summarize the dynamics of consumption, inflation, output and the interest rate :¹⁶

$$\hat{c}_t = \bar{\mathcal{D}}\mathbb{E}_t(\hat{c}_{t+1} - \hat{R}_t + \hat{\pi}_{t+1} + \hat{z}_{t+1})$$
(6)

$$\hat{\pi}_t = \beta \mathbb{E}_t \hat{\pi}_{t+1} + \bar{\varphi} \mathbb{E}_t \left(\hat{g}_{t+1} - \hat{g}_t \right) + \bar{\kappa} \left[\left(\frac{1}{\eta} + 1 \right) \hat{y}_t - \hat{g}_t \right] + \bar{\lambda} \hat{v}_t \tag{7}$$

$$\hat{y}_t = \hat{c}_t + \hat{g}_t \tag{8}$$

$$\hat{R}_t = 0 \tag{9}$$

The coefficients entering equations 6 and 7 are functions of the following structural parameters: $\bar{D} = \frac{\beta}{\beta + \bar{\pi} z \delta \bar{c}}$; $\bar{\kappa} = \frac{(1-\nu)g \bar{c} \bar{y}^{1/\eta}}{\nu \phi (2\bar{\pi} - \pi_*) \bar{\pi}}$; $\bar{\phi} = \frac{\bar{\pi} - \pi_*}{2\bar{\pi} - \pi_*}$; and $\bar{\lambda} = \frac{1-\phi(\bar{\pi} - \pi_*)\bar{\pi}(1-\beta)}{\phi(2\bar{\pi} - \pi_*)\bar{\pi}}$, where ϕ and δ correspond to the cost of price adjustment and the marginal utility of bonds, these are the only structural parameters specific to each model. The common parameters include β , the household's discount factor; η , the Frisch labor supply elasticity; $1/(1-\nu)$, the steady-state markup; z, the long-run growth of the economy; 1-1/g, the steady-state share of government spending relative to output; and π_* , the central banks' inflation target. The rest of the terms correspond to the steady-state value of consumption (\bar{c}) , output (\bar{y}) , inflation $(\bar{\pi})$, technology growth (z), government spending (g). Variables with an over-line denote values in liquidity trap steady state.

We obtain equation 6 from log-linearizing the modified Euler equation 5. It resembles the dynamic IS relationship of the standard New Keynesian model but modified by the discount coefficient \overline{D} . Since $\delta > 0$, the discounting coefficient $\overline{D} < 1$. Discounting dampens the consumption response to changes in the ex-ante real interest rate. An increase in the preference for bonds, lower steady-state inflation, and lower long-run growth rate increase the discounting in the Euler equation conditional on $\delta > 0$. We

¹⁶We denote all liquidity trap steady state parameters by \bar{x} . Appendix D provides the derivation of the log-linearized equations.

introduce shocks to growth rate of technology, z_t , to replicate the real interest rate observed in Japan.

Equation 7 is the forward-looking Phillips curve that depends on expected inflation and marginal costs $((1/\eta + 1) \hat{y}_t - \hat{g}_t)$, the growth in government expenditure $(\hat{g}_{t+1} - \hat{g}_t)$ and the price-markup shock \hat{v}_t . To generate this relationship we assume quadratic adjustment costs in price setting (Rotemberg, 1982). The growth in government expenditure appears in this equation because of we log-linearized the equation away from the targeted-inflation steady state.

Equation 8 is the resource constraint of the economy that specifies a time-varying wedge between consumption and output, corresponding to exogenous shocks in government spending. Equation 9 indicates that the economy operates under an interest rate peg. We can derive this equation from any policy rule in which the central bank faces an effective lower bound constraint.¹⁷

Exogenous shocks. There are three exogenous process in the model: (i) government expenditure g_t , (ii) the growth rate of productivity z_t , and (iii) changes in the inverse demand elasticity for intermediate goods, v_t , that translates into time-varying price markup. We assume that these exogenous components follow an AR(1) process around their deterministic mean (g, z, v), with persistence, ρ_g , ρ_z , ρ_v and innovations ϵ_g , ϵ_z , ϵ_v , that are normally distributed with mean zero and standard deviations σ_g , σ_z , σ_v , respectively.

3.3. Calibration

Table 1 summarizes the parameters that are common across models. We fix the discount factor β to 0.942 consistent with structural estimates of Gali and Gertler (1999). While this estimate is lower than the standard calibrated value of 0.99 in the literature, a low β is needed for the model to generate a positive natural interest rate in the presence of a bond premium. In studies that have estimated the discount rate using field and laboratory experiments, the estimates for β are dispersed but point to high

¹⁷As in Section 2, we assume that government runs a balanced budget every period. There is zero net supply of government bonds in the economy. We rebate the adjustment costs to the household to reduce the role of high adjustment costs in driving equilibrium dynamics.

discount rates. Surveys of these studies are conducted in Frederick, Loewenstein and O'Donoghue (2002, table 1), and Andersen, Harrison, Lau and Rutström (2014, table 3). Michaillat and Saez (2021) choose an annual discount rate of 43% from the median value of these estimates.

Other standard parameters include the Frisch labor supply elasticity fixed at 0.85 (Kuroda and Yamamoto, 2008). We set the (inverse) elasticity of demand for intermediate goods, ν , to 0.1 to generate a steady-state markup of 11%. Japan did not officially adopt an inflation target until 2013Q2, but the inflation rate averages 1.1% in the two decades before entering the ZLB. Thus we assume the central bank was pursuing an inflation target of 1% and use that target rate as the reference value for price adjustment($\Pi^* = 1.0025$).¹⁸ We determine the values of *z* such that the model matches the average output growth over the estimation sample. The steady-state value of government spending matches a consumption-output ratio of 58% in the Japanese data.¹⁹

Table 1: Fixed Parameters

β	η	ν	Π^*	g	$100 \ln(z)$
Discount factor	Inverse Frisch elasticity	Price s.s. markup	Inflation target	Government spending parameter	TFP growth rate
0.942	0.85	0.1	1.0025	1.72	0.56

The remaining parameters δ and ϕ are chosen to jointly match targets for the natural interest rate and average inflation in Japan.²⁰ For the natural rate, we adopt two different targets depending on the regime. Under secular stagnation, we choose an annual rate of -1.1%. This choice is based on two studies by Fujiwara, Iwasaki, Muto, Nishizaki and Sudo (2016) and Iiboshi, Shintani and Ueda (2018) that separately estimate a series for the natural rate of interest in Japan based on Laubach and Williams

¹⁸Our results are robust to choosing a zero inflation target as well.

¹⁹In our data, government expenditure is residually a combination of investment, net exports, and government spending. As an alternative, it is straightforward to make g_t in the model track actual government spending in the data by defining consumption appropriately. Results are available upon request.

²⁰It may be worth noting that multiple steady states at zero lower bound may coexist if the Phillips curve is sufficiently non-linear. Alternately, it may be possible to model the possibility of secular stagnation steady state coexisting with the full-employment steady state as in Eggertsson and Mehrotra (2015) with a sufficiently high inflation target. We do not explore those exercises here.

	δ Euler Equation	ф Adjustment cost	
Secular Stagnation	Wedge 0.1132	parameter 4825	
Expectations Trap	0.1088	2524	
	Natural Rate	Inflation	Output Gap
Secular Stagnation	-1.1	-1.06	-7.5
Expectations Trap	0	-1.06	-4.5

 Table 2: Calibrated Parameters

Notes: The table shows the parameter values of the model for the baseline calibration.

(2003). They find that the quarterly estimate was often -0.5% since the late 1990s and -2% at the lowest level. In contrast, we calibrate the expectations-trap steady state to imply an annualized long-run real interest rate of 0%. The calibration implies an inflation rate of -1.06% for both steady states, which is the average inflation rate in Japan over the period 1998Q4 – 2012Q4. Our calibration results in a somewhat larger value of the price adjustment parameter, ϕ , compared with econometric estimates of DSGE models for Japan (Iiboshi et al., 2018). Nonetheless, within the range of plausible estimates found in the literature—see Aruoba, Bocola and Schorfheide (2017) Finally, the implied output gap is close to the estimates of 5% in Hausman and Wieland (2014). Table 2 summarizes the parameters that are specific to each model.

3.4. Equilibrium dynamics

Similar to our analytical model in Section 2, the local dynamics of secular stagnation and expectations-trap are pretty different. The following proposition formalizes this result.

Proposition 5. (Local Determinacy) Assume $\beta < 1$. The system 6 - 9 is locally determinate if and only if $\frac{\tilde{\pi}z\delta_m\tilde{c}}{\beta} > \frac{1+\eta}{\eta(1-\beta)}\kappa_m$.

The secular stagnation steady state exhibits local-determinacy. This requires a sufficiently flat Phillips curve (low $\bar{\kappa}$), or high enough discounting (high δ).²¹ Our calibration satisfies this restriction. In contrast, an expectations-driven liquidity trap

²¹Definition 1 in Michaillat and Saez (2021) impose a similar restriction for obtaining a permanent fundamentals-driven ZLB episode.

has locally indeterminate dynamics, with low enough discounting or sufficiently steep Phillips curve.

Equilibrium indeterminacy carries two distinguishing features (Lubik and Schorfheide, 2004; Canova and Gambetti, 2010). One is the presence of additional endogenous variables affecting the dynamics of agents' forecast errors. Second is the possibility of extraneous innovations, known as sunspot shocks (ζ_t), that is not part of the original description of agents' optimization problems.

To characterize the multiplicity of equilibrium, we apply the methods in Bianchi and Nicolo (2021). This allows lagged inflation expectations $\mathbb{E}_{t-1}\hat{\pi}_t$ to enter as an additional state variable in agent's decision rules, and sunspot shocks to affect the inflation forecast error, $\eta_{\pi} \equiv (\hat{\pi} - \mathbb{E}_{t-1}\hat{\pi}_t) = \zeta_t$. We estimate the correlation between sunspot and structural innovations (ϵ_g , ϵ_z , ϵ_v), thus we select the best-fitting equilibrium based on observed Japanese data.

3.5. Data and Estimation

Conditional on our calibration of steady state parameters, we are left to estimate the vector of parameters $\theta = [\rho_g, \rho_z, \rho_v, \sigma_g, \sigma_z, \sigma_v]'$ for the secular stagnation model. For the expectations-trap model, in addition to the parameters in θ , we estimate the standard deviation of the sunspot shock and the correlation between the structural and the sunspot shocks, denoted by $\rho_{x,\zeta}$, for $x = \{z, g, v\}$. Because our model is linear we can construct the true-likelihood and use a standard Bayesian approach to estimate the parameters of the model. We obtain draws from the posterior distribution by a single-block random walk Metropolis–Hastings (RWMH) algorithm (An and Schorfheide, 2007). Appendix D reports the prior distribution of parameters.

Data. For parameter estimation, we use quarterly data on output growth, consumption growth, and GDP deflator-based inflation rate in Japan during the period 1998:Q1 to 2012:Q4.²² We focus on this sample period for two reasons. First, from 1995 to 1998 the Bank of Japan (BOJ) held the monetary policy rate at 0.5%, while struggling to

²²Our findings are robust to using data from 1998:Q1-2020:Q1 in estimation. We use the longer sample for our assessment of the mechanism in section 5.

boost the economy amidst turmoil in domestic and international financial markets (Ito and Mishkin, 2004). We start our analysis in 1998 to parallel the assumption in our model that the economy starts at the ZLB and agents expect near-zero interest rates for a prolonged period. The BOJ lowered its policy rate to zero in the first quarter of 1999, and it remained between 0% and 0.5%. Consequently, we consider the economy to be at the ZLB for the entire period. Second, in 2013, the BOJ introduced a new monetary policy program that included an explicit inflation target, asset and bond purchase programs as well as considering negative nominal interest rates (Gertler, 2017). None of these policies are explicitly modeled in our framework.

Measurement. To match the model to the data, we construct model implied output (Δy_t^o) , consumption growth (Δc_t^o) , as quarter-on-quarter percentages, and inflation measured in annualized percentages (π_t^A) . We link the observed data series to the model counterparts through the following system of measurement equations:

$$\Delta y_t^o = 100 \log(z) + 100 \left(\hat{y}_t - \hat{y}_{t-1} + \hat{z}_t \right)$$
$$\Delta c_t^o = 100 \log(z) + 100 \left(\hat{c}_t - \hat{c}_{t-1} + \hat{z}_t \right)$$
$$\pi_t^o = 400 \log(\bar{\pi}) + 400 \hat{\pi}_t$$

4. **Results**

4.1. Estimated parameters

Table 3 summarizes the estimated posterior distribution of parameters that fit the respective model to Japan's output, consumption, and inflation data in our sample. The marginal prior and posterior distributions for the estimated parameters are tabulated in the appendix. The posterior estimates for the common parameters are remarkably similar across model specifications. For the expectation traps model, the standard deviation of the sunspot shock is statistically different from zero and with a magnitude similar to that of the technology shock. In this specification, the estimated correlation between the fundamental and sunspot shocks varies substantially. The data favors a robust positive correlation between markup and sunspot shocks while picking up a

small correlation of the sunspot shock with the other two fundamental shocks.

Parameters	Description	\mathcal{M}_b : Expectations Trap	\mathcal{M}_s : SecularStagnation
ρ_{q}	Persistence gov. spending shock	0.9562	0.8620
10	0 1 0	[0.9197, 0.9947]	[0.7990, 0.9338]
$ ho_{ u}$	Persistence markup shock	0.1571	0.1433
	_	[0.0334, 0.2718]	[0.0387, 0.2465]
$ ho_z$	Persistence technology. growth shock	0.7358	0.7824
		[0.5093, 0.9443]	[0.5873, 0.9383]
σ_{g}	Std dev. gov. spending shock	0.0045	0.0045
0		[0.0039, 0.0051]	[0.0039, 0.0051]
$\sigma_{ u}$	Std dev. markup shock	0.0172	0.0026
		[0.0021, 0.0026]	[0.0021, 0.0030]
σ_z	Std dev. markup shock	0.0040	0.0037
		[0.0015, 0.0066]	[0.0014, 0.0065]
		Sunspot shock parameters	
σ_{ζ}		0.0030	-
0		[0.0026, 0.0035]	-
$ ho(\epsilon_z,\epsilon_\zeta)$		-0.0476	-
-		[-0.1870, 0.0999]	-
$ ho(\epsilon_ u,\epsilon_\zeta)$		0.9559	-
<i>.</i>		[0.9263, 0.9853]	-
$ ho(\epsilon_g,\epsilon_\zeta)$		-0.2130	-
		[-0.2831, -0.1410]	
$\log\left[p\left(Y^{T}\right)\right]$	Log-data density	-279.80	-284.80

 Table 3: Posterior DSGE estimates

Notes: The estimation sample is 1998:Q1 - 2012:Q4. We use $Y^T = [y_1, \ldots, y_T]$ to denote all the available data in our sample. For each model we report posterior means and 90% highest posterior density intervals in square brackets. All posterior statistics are based based on the last 25,000 draws from a RWMH algorithm, after discarding the first 25,000 draws.

4.2. Model Fit

We assess model fit using log-data density comparisons. The advantage of using logdata densities is that it intrinsically penalize the likelihood function for the presence of additional parameters. This approach is commonly used in the DSGE literature for model comparison (Lubik and Schorfheide, 2004; Justiniano and Primiceri, 2008; Cúrdia, Ferrero, Ng and Tambalotti, 2015; Ascari, Bonomolo and Lopes, 2019).

The last row in Table 3 shows that the log-data density favors the expectations-trap

hypothesis in terms of overall fit. To gauge the difference in fit, we construct Bayes factors of expectations-traps relative to secular stagnation, $\mathcal{F} = p(Y^T | \mathcal{M}_b) / p(Y^T | \mathcal{M}_s)$. As test statistic we compute $2 \times \log(\mathcal{F})$ because it resembles the familiar likelihood-ratio test. In our estimation, we find that this test statistic is equal to 10, implying "strong" evidence in favor of the expectations-trap hypothesis over secular stagnation according to standard criteria (Kass and Raftery, 1995).²³

In our application, the sunspot shock and the correlation parameters necessary to select equilibria in the expectations-trap model come at a cost from the perspective the log-data density.²⁴ Nonetheless, one may still be concerned that the expectations-trap model always "edges-over" secular stagnation because of the multiplicity of equilibria. To allay this concern, we conduct an exercise where we simulate data from the secular stagnation model using the parameters in Table 3. Then, we re-estimate both models on simulated data and conduct a model comparison. We find that $2 \times \log(\mathcal{F})$ is equal to -28, which indicates that when data comes from secular stagnation, our estimation procedure finds "very strong" evidence in its favor.

4.3. Impulse Response Functions

We now illustrate the difference in dynamics of expectation traps and secular stagnation through impulse responses. Figure 2 shows the output and the inflation impulse response functions after a one-time unanticipated shock to government expenditure, aggregate productivity growth, and price-markups.

For the secular stagnation model, shown by red dashed lines, all structural shocks lead to a transitory increase in inflation. With nominal interest rates at the ZLB, higher inflation translates into lower real interest rates and higher aggregate demand through the inter-temporal substitution channel. Thus, all shocks induce a positive conditional correlation between inflation and output.

The dynamics under expectations-trap are depicted with solid blue lines. Temporary

²³According to Kass and Raftery (1995), values of $2 \times \log(\mathcal{F})$ above 10 can be considered very strong evidence in favor of model 1. Values between 6 and 10 represent strong evidence, between 2 and 6 positive evidence, while values below 2 are "not worth more than a bare mention."

²⁴See the discussion in footnote 11 of Lubik and Schorfheide (2004).

increases in government spending or technology growth lower inflation. To understand how equilibrium indeterminacy affects the impact response of inflation, consider the expectation error $\eta_t = \hat{\pi}_t - \mathbb{E}_{t-1}\hat{\pi}_t$. Before any shock realizes, at time t = 0, the economy is at steady state with $\pi_0 = \bar{\pi}$ and $\mathbb{E}_0\hat{\pi}_0 = 0$. Then, $\hat{\pi}_1 = \eta_1$, which in turn depends on the correlation of the sunspot and the structural shocks. If structural shocks have a negative correlation with η , inflation will fall. At the zero lower bound, the decline in inflation raises the real interest rate and weighs on aggregate demand.

Technology shocks also have a negative correlation with sunspot shocks. Hence, a positive technology shock leads to an initial decline in inflation. However, a higher real interest rate does not fully offset the effects of higher future productivity and output increase on impact. Nevertheless, output declines as deflationary expectations set in and the real interest rate rises due to lower inflation.

Price-markup shocks increase inflation in about the same magnitude in both models. However, in the expectations-trap model, inflation rises on impact due to the correlation with the sunspot shock. After the initial jump, expected inflation declines as the transitory increase in realized inflation cannot persist in the expectations-trap steady state. Lower inflation expectations depress aggregate demand and generate a negative (conditional) correlation between inflation and output.

4.4. Expectations trap or secular stagnation?

We now compare the relative importance of the two competing hypotheses in explaining the persistent liquidity trap episode in Japan. We use static prediction pools, as in Geweke and Amissano (2011) and Del Negro et al. (2016), that rely on predictive densities to construct recursive estimates of model weights. These time-varying model weights can be interpreted as a policymaker's views on the most relevant model using the information available in real-time.

We consider a policymaker that has access to the sequence of one-period-ahead predictive densities $p(y_t|y_{1:t-1}, \mathcal{M}_s)$ under secular stagnation and $p(y_t|y_{1:t-1}, \mathcal{M}_b)$ under the expectations-trap hypothesis.²⁵ We are interested in constructing an estimate

²⁵The predictive density is constructed sampling from the posterior distribution of the DSGE



Figure 2: Impulse Responses: Expectations Trap vs Secular Stagnation

Notes: Impulse responses to one standard deviation shocks. All responses are computed at the posterior mean of the estimated parameters. The blue solid line corresponds to the expectations-driven traps model. The red dashed line corresponds to the secular stagnation model.

of the model weight, λ , that pools the information of each individual model:

$$p(y_t|\lambda, \mathcal{P}) = \lambda p(y_t|y_{1:t-1}, \mathcal{M}_b) + (1-\lambda)p(y_t|y_{1:t-1}, \mathcal{M}_s), \quad 0 \le \lambda \le 1$$
(10)

Where $p(y_t|\lambda, \mathcal{P})$ is the predictive density obtained by pooling the two competing models for a given weight λ and pool $\mathcal{P} = \{\mathcal{M}_b, \mathcal{M}_s\}$. The policymaker is Bayesian and has a prior density $p(\lambda|\mathcal{P})$ of the weight assigned to each model in the pool. The posterior distribution of the model weights, $p(\lambda|\mathcal{I}_t^{\mathcal{P}}, \mathcal{P})$, can be updated recursively

parameters of the baseline model of Section 4 and averaging the predictive densities across draws.

conditional on the information available to the pool in the previous period $\mathcal{I}_{t-1}^{\mathcal{P}}$:

$$p(\lambda | \mathcal{I}_t^{\mathcal{P}}, \mathcal{P}) \propto p(y_t | \lambda, \mathcal{P}) p(\lambda | \mathcal{I}_{t-1}^{\mathcal{P}}, \mathcal{P})$$
(11)

We estimate the posterior distribution in equation 11 recursively, starting in 1998:Q1. The estimated model weights are shown in Figure 3 together with posterior credible sets to capture model and parameter uncertainty. The Japanese data imply roughly similar weights on both models in the early part of the sample and through the early 2000s. Afterward, the data lean in favor of the specification \mathcal{M}_b , indicating a better fit of the expectations-trap hypothesis. Uncertainty about the model weight's posterior distribution is substantial but decreases later in the sample as more information favoring the expectations-trap model accumulates. From 2015, the data put at least 90% weight on the expectations-trap hypothesis as the best-fitting explanation.





Notes: The solid black line is the posterior mean of λ estimated recursively over the period 1998:Q1-2020:Q1. The shaded areas correspond to the 90 percent credible set of the posterior distribution.

5. Inspecting the Mechanism

As a result of local-indeterminacy, inflation in the expectations-trap model is free to jump away from its steady state in response to shocks. In section 4, we introduced i.i.d. sunspot shocks as exogenous shifters of inflation's expectational errors to select among the multiple self-confirming equilibria. Moreover, we allowed sunspot shocks to be correlated with structural shocks in the model. We now investigate why sunspot shocks matter? and why the correlation with structural shocks is essential for our results? We find that equilibrium indeterminacy relaxes the tight co-movement between inflation and output that afflicts the secular stagnation model.

5.1. Why Sunspots?

We construct the *Minimal State Variable* (MSV) solution corresponding to the expectationstrap model. This concept is common to select among solutions in models with equilibrium indeterminacy (Aruoba et al., 2018; Lansing, 2019). The idea is to restrict inflation and output dynamics to functions of the vector of fundamental state variables. In our model, the MSV criterion implies that the expectational error of inflation is determined by the exogenous disturbances \hat{z} , \hat{g} , \hat{v} and will not respond to other endogenous variables nor i.i.d. sunspot shocks. The following proposition formalizes the MSV solution concept and derives an analytical expression for our application:

Proposition 6. (MSV Solution). Let $X = \{g, v, z\}'$ collect all fundamental state variables of the model, and let *a* and *b* be vectors of unknown coefficients. Under the MSV criterion, a solution of the following form exists:

$$\hat{y}(\mathbb{X}) = a_1^j \hat{z}_t + a_2^j \hat{g}_t + a_3^j \hat{v}_t; \quad \hat{\pi}(\mathbb{X}) = b_1^j \hat{z}_t + b_2^j \hat{g}_t + b_3^j \hat{v}_t$$

for $j \in \{\text{Secular Stagnation, Expectations Trap}\}$. The coefficients (a_i^j, b_i^j) are reported in Appendix D.5.

To illustrate the MSV solution, figure 4 shows the impulse response of output and inflation to a markup shock. Naturally, for the secular stagnation model, the IRFs are identical to those shown in Figure 2. For the expectations-trap model, the MSV criterion rules out sunspots. The figure clarifies that the MSV solution is different from assuming a zero correlation between sunspot and structural shocks. The latter restricts inflation's forecast errors to zero. Thus inflation does not jump in response to structural shocks. In contrast, the MSV solution induces a contemporaneous correlation between inflation and output in response to structural shocks.

Under the MSV solution, the conditional correlation between inflation and output in the expectations-trap model is positive and similar to that in the secular stagnation model. This result shows that equilibrium selection directly affects the response of inflation to fundamental shocks and transmits through the economy through the inter-temporal substitution channel.

Price-markup shock increases realized as well as expected inflation. At the ZLB, a lower real interest stimulates aggregate demand. The following proposition analytically proves that this positive correlation result holds for all fundamental shocks in our model.



Figure 4: Impulse Response Functions: MSV solutions

Notes: Impulse responses to one standard deviation shock to price markups. All responses are computed at the posterior mean of the estimated parameters. The red solid line corresponds to inflation. The blue dashed line corresponds to output.

Proposition 7. (Positive Correlation). Consider the MSV solution for BSGU and Secular Stagnation models. Output and inflation are positively correlated conditional on shocks to TFP growth rate \hat{z}_t and price-markup \hat{v}_t . If $\bar{\kappa} < \frac{\bar{\pi}z\bar{c}\delta(1-\beta)}{\bar{R}\beta}$, the unconditional correlation of output growth and inflation is positive.

The intuition behind proposition 7 comes from the simple AS-AD graphs in section 2. Price-markup shocks and technology growth shocks shift only one schedule simultaneously– either the Phillips curve or the Euler equation. These shifters unequivocally induce a positive correlation between inflation and output. As long as the Phillips curve is sufficiently flat (low enough $\bar{\kappa}$ relative to other structural parameters), the government spending shock also induces a positive correlation between inflation and output. This restriction is satisfied by the parameters in our empirical exercise. Consequently, Proposition 7 implies that the expectations-trap model under the MSV criterion and the secular stagnation model yield qualitatively similar predictions.

The quantitative consequence of this proposition is that the likelihood of these two models is similar, thus making it challenging to identify the best-fitting model from the data. As we show in the Appendix D.5, the mean of the posterior distribution model weights $p(\lambda|.)$, is essentially constant at 50%. Our result shows that the equilibrium-multiplicity of solutions is not only theoretically relevant (Cochrane, 2011) but also quantitatively important.

5.2. Which Equilibrium?

The correlation between sunspot and structural shocks is crucial because it characterizes all admissible solutions under indeterminacy (Bianchi and Nicolo, 2021). Moreover, it helps discipline equilibrium selection using data.²⁶ Hence it is easy to study the equilibrium path that generates the quantitative success of the expectations-trap model by examining the correlation structure of the estimated model.

To understand which of the multiple equilibrium paths plays a role in discriminating between expectation traps and secular stagnation, we re-estimate the prediction pool under four restrictions on the correlation between the sunspot and fundamental shocks. Figure 5 displays the estimated time-varying model weights. Panel (a) sets to zero the correlation between the sunspot and productivity shocks. Panel (b) sets the correlation between the sunspot and the government expenditure shock to zero. Panel (c) sets the correlation of the sunspot shock and markup shock to zero. Lastly, panel (d) sets all the correlations to zero.

²⁶We focus on all linear rational expectations solutions. It is also possible to construct non-linear rational expectations solutions under indeterminacy—see Ascari et al. (2019).

When price-markups and sunspot shocks are uncorrelated as in panels (c) and (d), secular stagnation explains the Japanese experience better. Conversely, when the correlation between sunspot with productivity or government spending shocks is zero, we obtain results similar to our baseline specification, with expectation traps as the more likely explanation for the Japanese experience. We infer from these results that the correlation between price markups and sunspots shocks is crucial for the expectations trap hypothesis because it allows the model to generate a negative correlation between inflation and output. We elaborate on this empirical correlation in Section 5.3. This finding echoes the evidence presented in Wieland (2019) who shows that the oil supply shocks in Japan, which are equivalent to price markup shocks in our model, generate a negative correlation between inflation and output at the ZLB.





Notes: The solid black line is the posterior mean of λ estimated recursively over the period 1998:Q1-2020:Q1. The shaded areas correspond to the 90 percent credible set of the posterior distribution.

5.3. Inflation-Output Correlation

Now we turn to data moments that favor the expectations-trap hypothesis in our application. Our quantitative result is related to the ability of each model to generate an unconditional correlation between inflation and output that is consistent with the data.

Figure 6 shows the range of theoretical moments implied by the posterior parameter distribution of the expectation traps and secular stagnation model. The left panel shows the correlation between inflation and output growth. The right panel shows the volatility of inflation relative to the volatility of output growth. The blue shaded areas correspond to the theoretical range of moments generated by each model. The red dots in the figure represent the same moments in the Japanese data used in estimation.

The left panel shows the critical mechanism at play. The expectations-trap model can generate an unconditional negative correlation between inflation and output consistent with observed data. In contrast, the secular stagnation model cannot.²⁷ This result is the direct consequence of the conditional moments documented in the impulse response of Figure 2. The right panel shows that the expectations-trap model also generates relative volatility of inflation closer to the data. The tight co-movement of inflation and output moderates this relative volatility for the secular stagnation model.

Our results suggest that the fit of the secular stagnation hypothesis is impaired because the model has a restrictive set of exogenous shocks and cannot generate the empirical correlation of inflation and output. It is possible to relax model misspecification by allowing the correlation of fundamental shocks in the secular stagnation model, thus generating a negative inflation-output correlation. We do not see a clear economic interpretation to pursue such an approach. In contrast, in the expectations-trap model, the correlation between sunspot and fundamental shocks indexes an equilibrium as discussed in Bianchi and Nicolo (2021). Instead, in the next section, we show that relaxing misspecification through a more elaborate model structure does not overturn

²⁷Datta, Johannsen, Kwon and Vigfusson (2021) document a positive correlation between oil and equity prices in the U.S. post-2008. This measure is a proxy of the correlation between inflation and output in our model. We leave a formal quantitative assessment for the U.S. in our framework for future work.



Notes: Dots correspond to sample moments in Japan's data. Solid horizontal lines indicate medians of theoretical moments of the posterior distributions for parameter estimates and the boxes indicate 90% credible associated with the posterior distributions.

our main result.

6. Persistent Stagnation in a Medium-Scale DSGE Model

We extend our set up along the lines of Christiano, Eichenbaum and Evans (2005), Smets and Wouters (2007), or Justiniano, Primiceri and Tambalotti (2013). Our exercise confirms that our key finding of positive inflation-output correlation under secular stagnation also holds in a medium-scale DSGE model. This section briefly details the model structure and introduces new parameters used in our quantitative exercise. Relative to existing medium-scale DSGE models, the Euler equation central to our analysis is modified exactly in the same way as our three-equation setup in Section 3.

6.1. Model description

Compared to the model in Section 3, we introduce three changes. First, we allow for internal habits in consumption. Second, as in Gust, Herbst, López-Salido and Smith (2017), we incorporate a quadratic loss in household's utility due to wage adjustment and let monopolistically competitive households supply differentiated labor services, thus introducing nominal rigidities in wages. Third, we introduce
capital into production, with costly capital utilization and investment adjustment costs. Along each of these modifications we introduce three new structural shocks: timevarying wage-markups $(\frac{1}{1-\nu_{w,t}})$, a risk-premium shock (η_t^b) , and an investment-specific shock (μ_t) . A detailed description of the model is available in Appendix E.

We take most of the parameter values from Hirose (2020), who estimated a similar medium-scale model using Japanese data. The difference is that we approximate the model dynamics around the expectations-trap steady state. We also discuss how we re-calibrate some of the parameters to accommodate the secular stagnation hypothesis in our model.

Modified Euler-equation: Equation 12, is the consumption Euler equation, where λ_t represents the marginal utility of income and δ is the additive wedge induced by the preference for bonds. To be consistent with the secular stagnation calibration of the baseline model of Section 3, we set $\delta = 0.038$, such that the annualized natural rate of interest is -1.1%. With the presence of habits, marginal utility of income becomes a function of current and past consumption $\lambda_t = \frac{1}{c_t - \frac{hc_{t-1}}{z_t}} - \mathbb{E}_t \frac{h\beta}{z_{t+1}(c_{t+1} - \frac{hc_t}{z_{t+1}})}$, where the parameter *h* controls the persistence of the consumption habit, and z_t denotes the growth rate of TFP. We set h = 0.36 and $\beta = 0.942$.

$$\lambda_t = \beta \eta_t^b \mathbb{E}_t \left[\frac{\lambda_{t+1}}{z_{t+1}} \frac{1}{\Pi_{t+1}} \right] + \delta$$
(12)

Wage-setting: Equation 13 defines household's optimal wage-setting decisions and determines wage inflation (Π_t^w) . The parameter ψ_w defines the cost of wage adjustment, and $v_{w,t}$ represents time-varying wage markup parameter. The cost of wage adjustment depends on a reference inflation rate $\tilde{\Pi}_t^w = \bar{z} \bar{\Pi}^{1-\iota_w} (\exp(\epsilon_{z,t}) \Pi_t)^{\iota_w}$, which is a geometric average between steady-state wage inflation $\bar{\Pi}^\omega$ and price inflation Π_{t-1}^ω , with indexation weight given by the parameter $\iota_\omega = 0.30$. We set the disutility of hour worked $\omega = 0.50$ to hit the steady state labor hours in Hirose (2020), and the inverse of the Frisch labor elasticity $\frac{1}{\eta} = 2.3$. We re-calibrate the parameter ψ_w such that inflation

in the secular stagnation steady state equals an annualized rate of -1.06%.

$$\nu_{w,t}\psi_{w}\left[\frac{\Pi_{t}^{w}}{\tilde{\Pi}_{t-1}^{w}}-1\right]\frac{\Pi_{t}^{w}}{\tilde{\Pi}_{t-1}^{w}}=\nu_{w,t}\psi_{w}\beta\mathbb{E}_{t}\left[\frac{\Pi_{t+1}^{w}}{\tilde{\Pi}_{t}^{w}}-1\right]\frac{\Pi_{t+1}^{w}}{\tilde{\Pi}_{t}^{w}}+L_{t}\lambda_{t}\left[\omega\frac{L_{t}^{\frac{1}{\eta}}}{\lambda_{t}}-(1-\nu_{w,t})w_{t}\right]$$
(13)

Price of capital and investment: Equation 14 defines the value of an additional unit of capital relative to consumption, (q_t) , as a function of the expected marginal return to capital. The quarterly depreciation rate of capital is $\delta_k = 0.015$. To capture the cost of capital utilization we use $a(u_t) = (r^k / \sigma_a)(e^{(\sigma_a(u_t-1))} - 1)$, and set $\sigma_a = 2.246$.

$$q_{t} = \beta \mathbb{E}_{t} \left[\frac{\lambda_{t+1}}{\lambda_{t} z_{t+1}} \left(r_{t+1}^{K} u_{t+1} - a(u_{t+1}) + q_{t+1}(1 - \delta_{k}) \right) \right],$$
(14)

Equation 15 defines the optimal investment decision. To capture investment adjustment costs, we use the convex function $S(x_t) = \frac{\varphi_I}{2}(x_t - 1)^2$, and set the parameter $\varphi_I = 5.2$. The investment decision is also influenced by an exogenous investment specific shock μ_t . This shock changes the resource cost of transforming investment into installed capital.

$$q_{t}\mu_{t}\left[1-S\left(\frac{\mathbb{I}_{t}}{\mathbb{I}_{t-1}}\frac{z_{t}}{\bar{z}}\right)-S'\left(\frac{\mathbb{I}_{t}}{\mathbb{I}_{t-1}}\frac{z_{t}}{\bar{z}}\right)\frac{\mathbb{I}_{t}}{\mathbb{I}_{t-1}}\frac{z_{t}}{\bar{z}}\right]$$
$$+\beta\mathbb{E}_{t}\left[\mu_{t+1}\frac{\lambda_{t+1}}{\lambda_{t}}q_{t+1}\frac{z_{t+1}}{\bar{z}}\left(\frac{\mathbb{I}_{t+1}}{\mathbb{I}_{t}}\right)^{2}S'\left(\frac{\mathbb{I}_{t+1}}{\mathbb{I}_{t}}\frac{z_{t+1}}{\bar{z}}\right)\right]=1$$
(15)

Structural shocks: The six structural shocks driving the model economy are assumed to follow first order auto-regressive processes of the form $log(x_t) = (1 - \rho_x) log(x) + \rho_x log(x_{t-1}) + \sigma_x \epsilon_{x,t}$, with $\epsilon_{x,t} \sim N(0,1)$, and x denoting steady-state values, for $x_t = z_t, g_t, \eta_t^b, \mu_t, \nu_{p,t}, \nu_{w,t}$ (technology growth, government spending, risk-premium, investment-specific shock, price-markup and wage-markup respectively).

Equilibrium conditions We provide the full set of equilibrium conditions and additional parameter values in Appendix E.

6.2. Secular stagnation

We use the medium-scale DSGE model to generate two sets of moments: (i) impulse response functions of inflation and output to various shocks, and (ii) the model implied unconditional correlation between inflation and output.

Turning to the first set of moments, the impulse response functions in figure 7 show that, in our calibration, all the conditional correlations between inflation and output are positive. This result emerges from the positive co-movement of production and inflation in the absence of an active policy rule that responds to inflation deviations. In contrast, when the natural interest rate is non-negative, and nominal rates react to changes in inflation, higher markups or lower investment efficiency tend to reduce output and increase inflation. This mechanism is absent in the model linearized around the permanent liquidity trap.

Our second result is immediate from impulse response functions. The mediumscale model implies a positive correlation between inflation and output. The theoretical correlation for annualized inflation and output growth is 0.13, well within the range of implied theoretical correlations from the baseline model displayed in figure 6. The result confirms that secular stagnation is an unlikely candidate to explain the Japanese experience at the ZLB. The underlying reason is the limitation of the secular stagnation hypothesis to generate an inflation-output tradeoff in the absence of an active policy rule.

6.3. Expectation-traps

Finally, we briefly turn to the implications of the expectation-traps hypothesis in the context of the calibrated medium-scale DSGE model. Re-calibrating the model to an annualized natural interest rate of 0% and an annualized inflation rate of -1.06%, as in Section 3, we obtain a steady-state output gap of -4.3%. This calibration implies that the dynamics near the ZLB steady state are locally indeterminate. To select an equilibrium, we introduce a sunspot shock that is correlated with our six structural shocks, and calibrate the correlations between sunspot shocks and structural shocks



Figure 7: Secular Stagnation Dynamics in Medium-Scale NK Model

Notes: Impulse responses to one standard deviation shocks. The red solid lines correspond to inflation response. The blue dashed lines correspond to output responses.

using estimates from Hirose (2020).

In this case, the unconditional correlation between inflation and output growth is -0.12, consistent with our findings in Figure 6. Thus, as in Section 3, the expectations-trap model can fit a critical data moment of the Japanese data, and equilibrium indeterminacy remains essential to generate the negative correlation between inflation and output in a model of permanent liquidity traps.

7. CONCLUSION

In this paper, we developed a framework to formally model two hypotheses of stagnation: expectations-driven traps and secular stagnation. Our framework hinges on a modified Euler equation with discounting. We provide a tractable version of such modification that allows us to obtain analytical results. In this setting, we show the conditions under which expectations traps and secular stagnation emerge as steadystate equilibria of the model. We also show that the two hypotheses differ in the local determinacy properties. Because of contrasting predictions from traditional policies implemented at ZLB, we argue that there is a need for robust policies that can stimulate the economy regardless of the stagnation hypothesis. Our robust policies prescribe a flattening of the aggregate supply curve through price-indexation or minimum wage schemes.

We conduct an assessment of Japan's experience near the ZLB using a quantitative new-Keynesian model that embeds both hypotheses. We construct a time-varying probability of the relevant theory of stagnation. Our results show that the Japanese experience is consistent with the expectations-trap model. We find that the realtime assessment of both models has considerable uncertainty. We document that equilibrium indeterminacy of the expectations-trap model is central to account for empirical moments in Japanese data. In particular, the negative correlation between inflation and output growth. Our findings extend to a medium-scale model DSGE model of the Japanese economy.

References

- Andersen, Steffen, Glenn W Harrison, Morten I Lau, and E Elisabet Rutström. 2014. "Discounting behavior: A reconsideration." *European Economic Review*, 71: 15–33.
- An, Sungbae, and Frank Schorfheide. 2007. "Bayesian analysis of DSGE models." *Econometric reviews*, 26(2-4): 113–172.
- **Aruoba, Borağan, Luigi Bocola, and Frank Schorfheide.** 2017. "Assessing DSGE model nonlinearities." *Journal of Economic Dynamics and Control*, 83: 34–54.
- **Aruoba, S Borağan, Pablo Cuba-Borda, and Frank Schorfheide.** 2018. "Macroeconomic dynamics near the ZLB: A tale of two countries." *The Review of Economic Studies*, 85(1): 87–118.
- Ascari, Guido, and Jacopo Bonchi. 2020. "(Dis) Solving the Zero Lower Bound Equilibrium through Income Policy." LUISS School of European Political Economy WP Series.
- **Ascari, Guido, Paolo Bonomolo, and Hedibert F. Lopes.** 2019. "Walk on the Wild Side: Temporarily Unstable Paths and Multiplicative Sunspots." *American Economic Review*, 109(5): 1805–42.
- Auclert, Adrien, and Matthew Rognlie. 2018. "Inequality and Aggregate Demand." Working paper.

- **Benhabib, Jess, Stephanie Schmitt-Grohé, and Martín Uribe.** 2001. "The Perils of Taylor Rules." *Journal of Economic Theory*, 96(1): 40–69.
- Benhabib, Jess, Stephanie Schmitt-Grohé, and Martín Uribe. 2002. "Avoiding liquidity traps." *Journal of Political Economy*, 110(3): 535–563.
- Benigno, Gianluca, and Luca Fornaro. 2018. "Stagnation Traps." The Review of Economic Studies, 85(3): 1425–1470.
- **Bhattarai, Saroj, Gauti B Eggertsson, and Bulat Gafarov.** 2019. "Time consistency and the duration of government debt: A signalling theory of quantitative easing." Brown University manuscript.
- **Bianchi, Francesco, and Giovanni Nicolo.** 2021. "A Generalized Approach to Indeterminacy in Linear Rational Expectations Models." *Quantitative Economics (forthcoming)*.
- Bianchi, Francesco, and Leonardo Melosi. 2017. "Escaping the great recession." American Economic Review, 107(4): 1030–58.
- **Bilbiie, Florin Ovidiu.** 2018. "Neo-Fisherian Policies and Liquidity Traps." C.E.P.R. Discussion Papers CEPR Discussion Papers 13334.
- **Canova, Fabio, and Luca Gambetti.** 2010. "Do expectations matter? The great moderation revisited." *American Economic Journal: Macroeconomics*, 2(3): 183–205.
- **Christiano, L.J., M. Eichenbaum, and C.L. Evans.** 2005. "Nominal Rigidities and the Dynamic Effects of a Shock to Monetary Policy." *Journal of Political Economy*, 113(1).
- **Cochrane, John H.** 2011. "Determinacy and Identification with Taylor Rules." *Journal of Political Economy*, 119(3): 565–615.
- **Cochrane, John H.** 2017. "The new-Keynesian liquidity trap." *Journal of Monetary Economics*, 92: 47–63.
- **Coyle, Philip, and Taisuke Nakata.** 2019. "Optimal inflation target with expectationsdriven liquidity traps."
- **Cúrdia, Vasco, Andrea Ferrero, Ging Cee Ng, and Andrea Tambalotti.** 2015. "Has US monetary policy tracked the efficient interest rate?" *Journal of Monetary Economics*, 70: 72–83.
- **Das, Mausumi.** 2003. "Optimal growth with decreasing marginal impatience." *Journal of Economic Dynamics and Control*, 27(10): 1881–1898.
- **Datta, Deepa D., Benjamin K. Johannsen, Hannah Kwon, and Robert J. Vigfusson.** 2021. "Oil, Equities, and the Zero Lower Bound." *American Economic Journal: Macroeconomics*, 13(2): 214–53.

- **Del Negro, Marco, Gauti B Eggertsson, Andrea Ferrero, and Nobuhiro Kiyotaki.** 2017. "The great escape? A quantitative evaluation of the Fed's non-standard policies." *The American Economic Review*.
- **Del Negro, Marco, Raiden Hasengawa, and Frank Schorfheide.** 2016. "Dynamic Prediction Pools: An Investigation of Financial Frictions and Forecasting Performance." *Journal of Econometrics*, 192(1): 391–405.
- **Eggertsson, Gauti, Andrea Ferrero, and Andrea Raffo.** 2014. "Can structural reforms help Europe?" *Journal of Monetary Economics*, 61: 2–22. Carnegie-Rochester-NYU Conference Series on "Fiscal Policy in the Presence of Debt Crises" held at the Stern School of Business, New York University on April 19-20, 2013.
- **Eggertsson, Gauti B.** 2011. "What fiscal policy is effective at zero interest rates?" In *NBER Macroeconomics Annual* 2010, *Volume* 25. 59–112. University of Chicago Press.
- **Eggertsson, Gauti B.** 2012. "Was the new deal contractionary?" *American Economic Review*, 102(1): 524–55.
- Eggertsson, Gauti B., and Neil R. Mehrotra. 2015. "A Model of Secular Stagnation." Mimeo, Brown University.
- **Eggertsson, Gauti B., Neil R. Mehrotra, and Jacob A. Robbins.** 2019. "A Model of Secular Stagnation: Theory and Quantitative Evaluation." *American Economic Journal: Macroeconomics*, 11(1): 1–48.
- Eggertsson, Gauti B., Neil R. Mehrotra, Sanjay R. Singh, and Lawrence H. Summers. 2016. "A contagious malady? Open economy dimensions of secular stagnation." *IMF Economic Review*, 64(4): 581–634.
- **Epstein, Larry G.** 1983. "Stationary cardinal utility and optimal growth under uncertainty." *Journal of Economic Theory*, 31(1): 133–152.
- **Epstein, Larry G, and J Allan Hynes.** 1983. "The rate of time preference and dynamic economic analysis." *Journal of Political Economy*, 91(4): 611–635.
- **Feenstra, Robert C.** 1986. "Functional equivalence between liquidity costs and the utility of money." *Journal of Monetary Economics*, 17(2): 271–291.
- **Fisher, Jonas DM.** 2015. "On the Structural Interpretation of the Smets Wouters "Risk Premium" Shock." *Journal of Money, Credit and Banking*, 47(2-3): 511–516.
- **Frederick, Shane, George Loewenstein, and Ted O'Donoghue.** 2002. "Time discounting and time preference: A critical review." *Journal of economic literature*, 40(2): 351– 401.

- **Fujiwara, Shigeaki, Yuto Iwasaki, Ichiro Muto, Kenji Nishizaki, and Nao Sudo.** 2016. "Supplementary Paper Series for the "Comprehensive Assessment" (2): Developments in the Natural Rate of Interest in Japan." Bank of Japan Review Series 16-E-12.
- **Galı, Jordi, and Mark Gertler.** 1999. "Inflation dynamics: A structural econometric analysis." *Journal of monetary Economics*, 44(2): 195–222.
- Garga, Vaishali, and Sanjay R Singh. 2021. "Output Hysteresis and Optimal Monetary Policy." *Journal of Monetary Economics*, 105: 871–886.
- **Gertler, Mark.** 2017. "Rethinking the Power of Forward Guidance: Lessons from Japan." National Bureau of Economic Research Working Paper 23707.
- **Geweke, John, and Gianni Amissano.** 2011. "Dynamic Prediction Pools: An Investigation of Financial Frictions and Forecasting Performance." *Journal of Econometrics*, 164(1): 130–141.
- **Glover, Andrew.** 2019. "Avoiding Liquidity Traps With Minimum Wages: Can Stability Justify Distortions?" Mimeo, Kansas City Fed.
- **Gust, Christopher, Edward Herbst, David López-Salido, and Matthew E Smith.** 2017. "The empirical implications of the interest-rate lower bound." *American Economic Review*, 107(7): 1971–2006.
- Hansen, Alvin H. 1939. "Economic progress and declining population growth." *American Economic Review*, 29(1): 1–15.
- Hausman, Joshua K, and Johannes F Wieland. 2014. "Abenomics: preliminary analysis and outlook." *Brookings Papers on Economic Activity*, 2014(1): 1–63.
- **Heathcote, Jonathan, and Fabrizio Perri.** 2018. "Wealth and Volatility." *The Review of Economic Studies*, 85(4).
- **Hirose, Yasuo.** 2020. "An Estimated DSGE Model with a Deflation Steady State." *Macroeconomic Dynamics*, 24(4): 1151–1185.
- **Iiboshi, Hirokuni, Mototsugu Shintani, and Kozo Ueda.** 2018. "Estimating a Nonlinear New Keynesian Model with a Zero Lower Bound for Japan."
- **Ito, Takatoshi, and Frederic S Mishkin.** 2004. "Two Decades of Japanese Monetary Policy and the Deflation Problem." National Bureau of Economic Research Working Paper 10878.
- Justiniano, Alejandro, and Giorgio E Primiceri. 2008. "The time-varying volatility of macroeconomic fluctuations." *American Economic Review*, 98(3): 604–41.

- Justiniano, Alejandro, Giorgio E Primiceri, and Andrea Tambalotti. 2013. "Is there a trade-off between inflation and output stabilization?" *American Economic Journal: Macroeconomics*, 5(2): 1–31.
- Kass, Robert E., and Adrian E. Raftery. 1995. "Bayes Factors." *Journal of the American Statistical Association*, 90(430): 773–795.
- Krishnamurthy, Arvind, and Annette Vissing-Jorgensen. 2012. "The aggregate demand for treasury debt." *Journal of Political Economy*, 120(2): 233–267.
- Kuroda, Sachiko, and Isamu Yamamoto. 2008. "Estimating Frisch labor supply elasticity in Japan." *Journal of the Japanese and International Economies*, 22(4): 566–585.
- **Lansing, Kevin J.** 2019. "Endogenous forecast switching near the zero lower bound." *Journal of Monetary Economics*.
- Laubach, Thomas, and John C Williams. 2003. "Measuring the natural rate of interest." *Review of Economics and Statistics*, 85(4): 1063–1070.
- Lubik, Thomas A, and Frank Schorfheide. 2004. "Testing for indeterminacy: an application to US monetary policy." *American Economic Review*, 94(1): 190–217.
- **McCallum, Bennett T.** 2004. "On the relationship between determinate and MSV solutions in linear RE models." *Economics Letters*, 84(1): 55 60.
- Mertens, Karel, and Morten O Ravn. 2014. "Fiscal policy in an expectations-driven liquidity trap." *The Review of Economic Studies*, 81(4): 1637–1667.
- Mertens, Thomas M, and John C Williams. 2021. "What to Expect from the Lower Bound on Interest Rates: Evidence from Derivatives Prices." *American Economic Review (forthcoming)*.
- **Michaillat, Pascal, and Emmanuel Saez.** 2021. "Resolving New Keynesian Anomalies with Wealth in the Utility Function." *The Review of Economics and Statistics*, 103(2): 197–215.
- Michau, Jean-Baptiste. 2018. "Secular stagnation: Theory and remedies." *Journal of Economic Theory*, 176: 552–618.
- Nakata, Taisuke. 2017. "Uncertainty at the zero lower bound." *American Economic Journal: Macroeconomics*, 9(3): 186–221.
- Nakata, Taisuke, and Sebastian Schmidt. 2019. "Conservatism and liquidity traps." *Journal of Monetary Economics*, 104: 37–47.
- Nakata, Taisuke, and Sebastian Schmidt. 2021. "Expectations-Driven Liquidity Traps: Implications for Monetary and Fiscal Policy." *American Economic Journal: Macroeconomics (forthcoming)*.

- **Ono, Yoshiyasu, and Katsunori Yamada.** 2018. "Difference or Ratio: Implications of Status Preference on Stagnation." *Australian Economic Papers*.
- Rotemberg, Julio J. 1982. "Sticky prices in the United States." *Journal of Political Economy*, 90(6): 1187–1211.
- **Schmidt, Sebastian.** 2016. "Lack of confidence, the zero lower bound, and the virtue of fiscal rules." *Journal of Economic Dynamics and Control*, 70: 36–53.
- Schmitt-Grohé, Stephanie, and Martín Uribe. 2003. "Closing small open economy models." *Journal of international Economics*, 61(1): 163–185.
- Schmitt-Grohé, Stephanie, and Martín Uribe. 2017. "Liquidity traps and jobless recoveries." *American Economic Journal: Macroeconomics*, 9(1): 165–204.
- Schmitt-Grohé, Stephanie, and Martín Uribe. Forthcoming. "Exchange Rates and Uncovered Interest Differentials: The Role of Permanent Monetary Shocks." *Journal of International Economics*.
- Schwarz, Gideon. 1978. "Estimating the dimension of a model." *The annals of statistics*, 461–464.
- **Smets, Frank, and Rafael Wouters.** 2007. "Shocks and frictions in US business cycles: A Bayesian DSGE approach." *American Economic Review*, 97(3): 586–606.
- **Summers, Lawrence H.** 2013. "Have we Entered an Age of Secular Stagnation." *IMF Fourteenth Annual Research Conference in Honor of Stanley Fischer, Washington, DC,* 63(1): 277–280.
- **Uribe, Martín.** 2021. "The neo-Fisher effect: Econometric evidence from empirical and optimizing models." *American Economic Journal: Macroeconomics*.
- **Uzawa, Hirofumi.** 1968. "Time preference, the consumption function, and optimum asset holdings." *Value, capital and growth: papers in honor of Sir John Hicks. The University of Edinburgh Press, Edinburgh*, 485–504.
- Wieland, Johannes F. 2019. "Are negative supply shocks expansionary at the zero lower bound?" *Journal of Political Economy*, 127(3): 973–1007.
- **Woodford, Michael.** 2003. *Interest and prices: Foundations of a theory of monetary policy.* Princeton University Press Princeton, NJ.

A. PROOFS FOR PROPOSITIONS IN SECTION 2

Proposition (Proposition 1: Targeted Steady State). Let $0 < \delta < \frac{1}{\beta}$. There exists a unique positive interest rate steady state with Y = 1, $\Pi = 1$ and $R = \frac{1}{\beta\delta} > 1$. It features output at efficient steady state, and inflation at the policy target. The equilibrium dynamics in this steady state's neighborhood are locally determinate.

Proof. The downward sloping portion of aggregate demand always goes through Y = 1, $\Pi = 1$. When $\delta < \frac{1}{\beta}$, $1 + r^* > 1$. The kink in the AD curve occurs at $\Pi_{kink} = \left(\frac{1}{(1+r^*)}\right)^{\frac{1}{\phi_{\pi}}} < 1$ and $Y_{AD,kink} = (1+r^*)^{1-\frac{1}{\phi_{\pi}}} > 1$. There always exists an intersection between the AS and the AD at Y = 1 and $\Pi = 1$. To show that there does not exist another steady state at positive interest rates, note the AS curve is linear and upward sloping. For $\Pi > 1$, $Y_{AD} < 1 < Y_{AS}$. And for $\Pi_{kink} \leq \Pi < 1$, $Y_{AD} > 1 > Y_{AS}$. There does not exist another steady state at positive nominal interest rate.

To prove local-determinacy, log-linearize the equilibrium conditions (1) - (3) around the unique non-stochastic steady state Y = 1, $\Pi = 1$ and $R = \frac{1}{\beta\delta} > 1$. The system of equations can be simplified to:

$$\hat{Y}_t = \frac{1+\kappa}{2+\phi_\pi\kappa} \mathbb{E}_t \hat{Y}_{t+1} + \hat{r}_t^n$$

where hat variables represent log-deviations from steady state. Given, $\kappa > 0$ and $\phi_{\pi} > 1$, this system satisfies Blanchard-Kahn conditions for determinacy of linear rational-expectations equilibria.

Proposition (Proposition 2: Expectations trap steady state). Let $0 < \delta < \frac{1}{\beta}$. For $\kappa > 1$ (i.e. $\alpha_p > 0.5$) there exist two steady states:

- 1. The targeted steady state with Y = 1, $\Pi = 1$ and $R = \frac{1}{\beta\delta} > 1$.
- 2. (*Expectations-driven* trap) A unique-ZLB steady state with $Y = \frac{1-\kappa}{\beta\delta-\kappa} < 1$, $\Pi = \frac{\beta\delta(1-\kappa)}{\beta\delta-\kappa} < 1$ and R = 1. The local dynamics in a neighborhood around the unemployment steady state are locally in-determinate.

When prices are rigid enough, i.e. $\kappa < 1$, there exists a unique steady state and it is the targeted inflation steady state. When prices are flexible $\alpha_p = 1$ ($\kappa \rightarrow \infty$), there always exist two steady states: a unique deflationary steady state with zero nominal interest rates and a unique targeted inflation steady state.

Proof. For
$$\kappa > 1$$
:

When $\delta < \frac{1}{\beta}$, $1 + r^* > 1$. The kink in the AD curve occurs at $\Pi_{kink} = \left(\frac{1}{(1+r^*)}\right)^{\frac{1}{\phi\pi}} < 1$ and

 $Y_{AD,kink} = (1 + r^*)^{1 - \frac{1}{\phi \pi}} > 1$. There always exists an intersection between the AS and the AD at Y = 1 and $\Pi = 1$. To show that there doesn't exist another steady state at positive interest rates, note that the AS curve is linear and upward sloping. For $\Pi > 1$, $Y_{AD} < 1 < Y_{AS}$. And for $\Pi_{kink} \leq \Pi < 1$, $Y_{AD} > 1 > Y_{AS}$. There does not exist another steady state at positive nominal interest rate. The proof for local-determinacy of this targeted steady state follows similar steps as in Proposition 1.

To prove that there exists a unique intersection at zero nominal interest rates, we note that AS and AD are linear for $\Pi < \Pi_{kink}$. When $\Pi = \Pi_{kink}$, $Y_{AD} > 1 > Y_{AS}$. And when $\Pi = 0$,

 $Y_{AD} = 0 < Y_{AS} = \frac{\kappa - 1}{\kappa} > 0$. This is because of the assumption that $\alpha_p > 0.5$. Hence, there exists a unique intersection at zero nominal interest rates. To prove local-indeterminacy, log-linearize the equilibrium conditions (1) - (3) around the unique non-stochastic steady state $Y_S = \frac{1-\kappa}{\beta\delta - \kappa} < 1$, $\Pi_S = \frac{\beta\delta(1-\kappa)}{\beta\delta - \kappa} < 1$ and $R_S = 1$. The system of equations can be simplified to:

$$\hat{Y}_t = \frac{1 + \kappa_b}{2} \mathbb{E}_t \hat{Y}_{t+1} + \hat{r}_t^n$$

where hat variables represent log-deviations from steady state and $\kappa_b \equiv \frac{\kappa Y_S}{\kappa Y_S + 1 - \kappa} > 1$ (because $1 - \kappa < 0$ and $\beta \delta < 1$). Hence, this system does not have a unique bounded rational expectations equilibrium.

<u>For $\kappa < 1$ </u>:

That there exists a unique non-ZLB steady state follows from Proposition 1. Remains to show that there does not exist a ZLB steady state. Note that, AD is linear and upward sloping when ZLB is binding and AS is always linear. Furthermore for $\Pi_{kink} \leq \Pi < 1$, $Y_{AD} > 1 > Y_{AS}$. And for $\Pi = 1 - \kappa$, $Y_{AS} = 0 < Y_{AD}$. Thus there does not exist a steady state with zero nominal interest rate.

Proposition (Proposition 3: Secular Stagnation). Let $\delta > \frac{1}{\beta}$ and $\kappa < 1$. There exists a unique steady state with $Y = \frac{1-\kappa}{\beta\delta-\kappa} < 1$, $\Pi = \frac{\beta\delta(1-\kappa)}{\beta\delta-\kappa} < 1$ and R = 1. It features output below the targeted steady state and deflation, caused by a permanently negative natural interest rate. The equilibrium dynamics in this steady state's neighborhood are locally determinate.

Proof. When $\delta > \frac{1}{\beta}$, $1 + r^* < 1$, thus, the kink in the AD occurs at $\Pi_{kink} > 1$, $Y_{AD,kink} < 1$. For $\Pi > \Pi_{kink}$, $Y_{AD} < 1 < Y_{AS}$. Thus, no steady state exists at positive nominal interest rates. When $\Pi = \Pi_{kink}$, $Y_{AD} < 1 < Y_{AS}$. For $\Pi < \Pi_{kink}$, AS and AD are both and downward sloping. At $\Pi = 1 - \kappa < 1$, $Y_{AS} = 0 < Y_{AD}$. Hence there exists a unique steady state of the economy with nominal rigidities.

To prove local-determinacy, log-linearize the equilibrium conditions (1) - (3) around the unique non-stochastic steady state $Y_S = \frac{1-\kappa}{\beta\delta-\kappa} < 1$, $\Pi_S = \frac{\beta\delta(1-\kappa)}{\beta\delta-\kappa} < 1$ and $R_S = 1$. The system of equations can be simplified to:

$$\hat{Y}_t = \frac{1 + \kappa_a}{2} \mathbb{E}_t \hat{Y}_{t+1} + \hat{r}_t^n$$

where hat variables represent log-deviations from steady state and $\kappa_a \equiv \frac{\kappa Y_S}{\kappa Y_S + 1 - \kappa} < 1$. Given, $0 < \kappa_a < 1$, this system satisfies Blanchard-Kahn conditions for determinacy of linear rational-expectations equilibria.

B. DERIVATION OF THE PRICE INDEXATION SCHEME

As shown in Section 2.2, the price index is given by

$$P_t^{\frac{\nu-1}{\nu}} = \alpha \left(p_t^* \right)^{\frac{\nu-1}{\nu}} + (1-\alpha) \left(p_t^n \right)^{\frac{\nu-1}{\nu}}$$

It can be rewritten as

$$1 = \alpha \left(\frac{p_t^*}{P_t}\right)^{\frac{\nu-1}{\nu}} + (1-\alpha) \left(\frac{p_t^n}{P_t}\right)^{\frac{\nu-1}{\nu}}$$
$$= \alpha \left(\frac{Y_t}{\bar{Y}}\right)^{\frac{\nu-1}{\nu}} + (1-\alpha) \left(\Gamma_t \frac{P_{t-1}}{P_t}\right)^{\frac{\nu-1}{\nu}}$$

If $\nu = 1/2$, we get:

$$1 = \alpha \left(\frac{Y_t}{\bar{Y}}\right)^{-1} + (1 - \alpha) \left(\Gamma_t \frac{P_{t-1}}{P_t}\right)^{-1}$$
$$(1 - \alpha) \left(\Gamma_t^{-1} \Pi_t - 1\right) = \alpha \left(\frac{Y_t - \bar{Y}}{Y_t}\right)$$

Define $\Gamma_t = \frac{P_t}{Y_t^{-1}(P_t - \lambda P_{t-1}) + P_{t-1}}$, to get

$$(\Pi_t - \lambda) = \frac{\alpha}{1 - \alpha} (Y_t - \bar{Y})$$

When $\lambda = 1$, the price Phillips curve simplifies to

$$(\Pi_t - 1) = \frac{\alpha}{1 - \alpha} (Y_t - \bar{Y})$$

 $\forall \lambda > \kappa > 1$, there does not exist expectations trap. An indexation to Γ_t of yesterday's price with $\lambda > \kappa$ can eliminate expectations trap.

Another indexation in price setting that we can assume is $\Gamma_t = Y_t$. This gives rise to the following relationship between gross inflation and output:

$$\Pi_t = (1+\kappa)Y_t - \kappa$$

where $\kappa = \frac{\alpha_p}{1-\alpha_p} > 0$ as before. With this Phillips curve, there always exist two steady states as long as $0 < \delta\beta < 1$ and $\kappa > 0$. We can analytically derive the steady states as in Proposition 2. A takeaway of our analysis is that appropriate price/wage indexation schemes can eliminate expectations trap. These can also be shown to improve the outcome in secular stagnation (as a corollary of paradox of flexibility).

A set of policies that are robust to the kind of stagnation can be framed from our analysis:

Corollary 1. (robust policies) A downwardly rigid price/wage indexation scheme can eliminate expectations trap while also improving welfare under secular stagnation.

C. QUANTITATIVE MODEL

This section describes the micro-foundations behind the log-linearized setup presented in Section 3.

C.1. Households

$$\max_{C_t(k),H_t,B_t(k)} \mathbb{E}_0\left[\sum_{t=0}^{\infty} \beta^t \left(\log\left(C_t(k)/A_t\right) - \omega \frac{H_t^{1+1/\eta}}{1+1/\eta} - \delta \frac{B_t}{A_t P_t}\right)\right],$$

subject to:

$$P_tC_t(k) + B_t(k) + T_t = W_tH_t + R_{t-1}B_{t-1}(k) + P_tD_t(k) + P_tSC_t,$$

The household derives utility from consumption C_t and from holding (real) stock of riskfree nominal bonds B_t , and disutility from hours worked H_t . The parameter ω scales the steady-state level of hours worked. The parameter δ regulates the marginal utility from holding bonds. The risk-free nominal bond pays a gross nominal interest rate R_t Each household supplies homogeneous labor services H_t in a competitive labor market taking the aggregate wage W_t as given. It collects interest payments on bond holdings, real profits D_t from intermediate good producers, pays lump sum taxes T_t , and receives payouts SC_t from trading a full set of state(k)-contingent securities.

Consumption decision. Let $\beta^t \lambda_t$ be the Lagrange multiplier on the household budget constraint, the first-order condition with respect to consumption and bond holdings are given by:

$$\left(\frac{C_t}{A_t}\right)^{-1} \frac{1}{A_t} = \beta R_t \left(\left(\frac{C_{t+1}}{A_{t+1}}\right)^{-1} \frac{1}{A_t}\right) \frac{P_t}{P_{t+1}} + \frac{\delta}{A_t}$$
(C.1)

Define:

$$Q_{t+1|t} = \frac{\lambda_{t+1} P_{t+1}}{\lambda_t P_t} \tag{C.2}$$

Using this definition, the first-order condition for bond holdings becomes:

$$1 = \beta \mathbb{E}_t \left[Q_{t+1|t} R_t \frac{P_t}{P_{t+1}} \right] + \delta \left(\frac{C_t}{A_t} \right)$$
(C.3)

The stochastic discount factor can be written as:

$$Q_{t+1|t} = \left(\frac{C_{t+1}/A_{t+1}}{C_t/A_t}\right)^{-1} \frac{A_t}{A_{t+1}} = \left(\frac{C_{t+1}/A_{t+1}}{C_t/A_t}\right)^{-1} \frac{1}{\gamma z_{t+1}}$$
(C.4)

Combining with the FOC for bond holdings we obtain the expression for the Euler equation:

$$1 = \beta \mathbb{E}_t \left[\left(\frac{C_{t+1}/A_{t+1}}{C_t/A_t} \right)^{-1} \frac{R_t}{\gamma z_{t+1}} \frac{P_t}{P_{t+1}} \right] + \delta \left(\frac{C_t}{A_t} \right)$$
(C.5)

Labor supply. Taking first-order conditions with respect to H_t yields

$$-\omega H_t^{1/\eta} + W_t \lambda_t = 0 \tag{C.6}$$

Combine with the FOC for consumption and defining the real wage as $w_t = \frac{W_t}{P_t}$ to obtain:

$$\frac{w_t}{A_t} = \omega H_t^{1/\eta} \left(\frac{C_t}{A_t}\right)^{-1} \tag{C.7}$$

C.2. Final Good Firms

The perfectly competitive, representative, final-good producing firm combines a continuum of intermediate goods indexed by $j \in [0, 1]$ using the technology:

$$Y_t = \left(\int_0^1 Y_t(j)^{1-\nu_t} dj\right)^{\frac{1}{1-\nu_t}}$$

Here $1/\nu_t > 1$ represents the elasticity of demand for each intermediate good. Profit maximization implies that the demand for intermediate goods is given by:

$$Y_t(j) = \left(\frac{P_t(j)}{P_t}\right)^{-1/\nu_t} Y_t.$$

where the price of the final good P_t is given by :

$$P_t = \left(\int_0^1 P_t(j)^{\frac{\nu_t - 1}{\nu_t}} dj\right)^{\frac{\nu_t}{\nu_t - 1}}$$

C.3. Intermediate Good Producers

Intermediate good *j* is produced by a monopolist who has access to the following production technology:

$$Y_t(j) = A_t H_t(j)$$
, with $A_t = A_{t-1} z_t$,

where A_t denotes the aggregate level of technology that is common to all firms, and z_t represents the stochastic (stationary) movements in TFP.

Intermediate good producers buy labor services $H_t(j)$ at a nominal price of W_t . Moreover, they face nominal rigidities in terms of price adjustment costs. These adjustment costs, expressed as a fraction of total output, are defined by the function $\Phi_p(.)$:

$$\Phi_p\left(\frac{P_t(j)}{P_{t-1}(j)}\right) = \frac{\phi}{2}\left(\frac{P_t(j)}{P_{t-1}(j)} - \Pi^*\right)^2 Y_t$$

where Π^* is the inflation rate in the targeted steady state. Taking as given nominal wages, final good prices, the demand schedule for intermediate products and technological constraints, firm *j* chooses its the price $P_t(j)$ to maximize the present value of future profits:

$$\max_{\{P_{t+s}(j)\}} \mathbb{E}_t \sum_{s=0}^{\infty} \beta^s Q_{t+s|t} \left(\frac{P_{t+s}(j)}{P_{t+s}} Y_{t+s}(j) - \Phi_p \left(\frac{P_{t+s}(j)}{P_{t+s-1}(j)} \right) Y_{t+s} - \frac{W_{t+s}Y_{t+s}(j)}{z_{t+s}P_{t+s}} \right) ,$$

subject to

$$Y_t(j) = \left(\frac{P_t(j)}{P_t}\right)^{-1/\nu_t} Y_t.$$

Price setting decision. Denoting $\mu_{t+s}\beta^s Q_{t+s|t}$ as the Lagrange multiplier associated with this constraint. The first-order condition with respect to $P_t(j)$ is given by

$$0 = \frac{A_t H_t(j)}{P_t} - \Phi' \left(\frac{P_t(j)}{P_{t-1}(j)} \right) \frac{A_t H_t(j)}{P_{t-1}(j)} - \frac{\mu_t}{\nu_t} \left(\frac{P_t(j)}{P_t} \right)^{-1/\nu_t - 1} \frac{Y_t}{P_t} + \beta \mathbb{E}_t \left[Q_{t+1|t} \Phi' \left(\frac{P_{t+1}(j)}{P_t(j)} \right) A_{t+1} H_{t+1}(j) \frac{P_{t+1}(j)}{P_t^2(j)} \right].$$

Labor demand. Taking first-order conditions with respect to $H_t(j)$ yields

$$\frac{W_t}{A_t P_t} = \frac{P_t(j)}{P_t} - \Phi\left(\frac{P_t(j)}{P_{t-1}(j)}\right) - \mu_t$$

Symmetric equilibrium. Let $P_t(j) = P_t \forall j$, the firm's equilibrium condition become:

$$\begin{aligned} \frac{\mu_t}{\nu_t} + \Phi_p'\left(\frac{P_t}{P_{t-1}}\right)\frac{P_t}{P_{t-1}} - 1 &= \beta \mathbb{E}_t \left[Q_{t+1|t}\frac{P_{t+1}}{P_t}\Phi'\left(\frac{P_{t+1}}{P_t}\right)\frac{Y_{t+1}}{Y_t}\right]\\ \mu_t &= 1 - \Phi\left(\frac{P_t}{P_{t-1}}\right) - \frac{w_t}{A_t}\end{aligned}$$

Combine both equilibrium conditions with the optimal labor supply condition from the household problem to obtain the FOC determining optimal price-setting:

$$(1 - \nu_t) - \omega H_t^{1/\eta} \left(\frac{C_t}{A_t}\right)^{-1} + \nu_t \Phi' \left(\frac{P_t}{P_{t-1}}\right) \frac{P_t}{P_{t-1}} = \nu_t \beta \mathbb{E}_t \left[Q_{t+1|t} \frac{P_{t+1}}{P_t} \Phi' \left(\frac{P_{t+1}}{P_t}\right) \frac{Y_{t+1}}{Y_t} \right]$$
(C.8)

Government Policies

The desired policy rate is set according to the following rule:

$$R_t^* = \left[r \Pi^* \left(\frac{\Pi_t}{\Pi^*} \right)^{\psi_1} \right],$$

Here *r* is the steady-state real interest rate, Π_t is the gross inflation rate defined as Π_t , and Π^* is the target inflation rate, which in equilibrium coincides with the steady state inflation rate. The actual policy rate relevant for agents decisions is subject to the zero lower bound constraint:

$$R_t = \max\{1, R_t^*\}$$

The government levies a lump-sum tax (subsidy) to finance any shortfalls in government revenues (or to rebate any surplus). The government's budget constraint is given by:

$$P_tG_t + R_{t-1}B_{t-1} = T_t + B_t,$$

where $G_t = \left(1 - \frac{1}{g_t}\right) Y_t$ is the government expenditure.

C.4. Resource constraint

We assume that the price adjustment costs are rebated back to the household in lump-sum fashion as part of the government transfers.²⁸ Hence, the market-clearing resource constraint is given by:

$$C_t + G_t = Y_t$$

Finally, we assume nominal bonds are in net zero supply

$$B_t = 0$$

D. DSGE Solution and Estimation

This section describes how we obtain the equations of the log-linearized model of Section 3, as well as proofs Propositions 5, 6 and 7.

²⁸An analogous interpretation would be to consider these costs as mental accounting costs for the firms or model these in the utility function of the representative agent. This assumption allows us to avoid unnatural results commonly associated with resource costs modeled in terms of output.

D.1. Log-linearized model

With a slight abuse of notation, we can write the system of equilibrium conditions as follow:

$$1 = \frac{\beta R}{\pi z} e^{-(c_{t+1}-c_t)+R_t-\pi_{t+1}-z_{t+1}} + \delta c e^{c_t}$$

$$(1 - v e^{v_t}) + v \phi e^{v_t} (\pi e^{\pi_t} - \pi_*) \pi e^{\pi_t} = \omega c y^{1/\eta} e^{c_t + (1/\eta)y_t}$$

$$+ v \phi \beta \pi e^{v_t} \left[e^{-(c_{t+1}-c_t)+y_{t+1}-y_t+\pi_{t+1}} (\pi e^{\pi_{t+1}} - \pi_*) \right]$$

$$c e^{c_t} = \frac{y}{g} e^{y_t - g_t}$$

Linearization around an arbitrary (R, π) point yields:

$$\begin{split} c_{t} &= \frac{R\beta}{(R\beta + \pi z \delta c)} c_{t+1} - \frac{R\beta}{(R\beta + \pi z \delta c)} \left(R_{t} - \pi_{t+1} - z_{t+1} \right) \\ \pi_{t} &= \beta \frac{(\pi - \pi_{*})}{2\pi - \pi_{*}} \left[(c_{t+1} - c_{t}) + y_{t+1} - y_{t} \right] + \beta \pi_{t+1} + \left(\frac{\nu - (1 - \beta)\nu\phi\pi(\pi - \pi_{*})}{\nu\phi\pi(2\pi - \pi_{*})} \right) \nu_{t} \\ &+ \left(\frac{(1 - \nu)y^{1 + 1/\eta}}{\nu\phi\pi(2\pi - \pi_{*})} \right) \left(c_{t} + \frac{1}{\eta} y_{t} \right) \\ c_{t} &= y_{t} - g_{t} \end{split}$$

D.2. Full employment

Around the full employment steady state we have $\pi = \pi_*$ and $R = \pi_* r_0$. In our calibration for the full employment steady state we have $r_0 = \frac{z(1-\delta g^{-1})}{\beta} = \exp(1/400)$ and $\pi_* = \exp(1/400)$. Moreover, we choose ω to normalize the full employment level of output to y = 1.

$$c_{t} = \mathcal{D}c_{t+1} - \mathcal{D}\left(R_{t} - \pi_{t+1} - z_{t+1}\right)$$

$$\pi_{t} = \beta \pi_{t+1} + \lambda \nu_{t} + \kappa c_{t} + \left(\frac{\kappa}{\eta}\right) y_{t}$$

$$c_{t} = y_{t} - g_{t}$$

$$(D.1)$$

Where, $\mathcal{D} = \frac{R\beta}{(R\beta + \pi z \delta c)}$, $\lambda = \left(\frac{\nu}{\nu \phi \pi^2}\right)$, and $\kappa = \left(\frac{1-\nu}{\nu \phi \pi^2}\right)$.

D.3. Permanent Liquidity Trap

When the economy is at a permanent liquidity trap, we have R = 1. We denote by \bar{x} the steady state values corresponding to the liquidity trap steady state.

$$\begin{split} c_{t} &= \frac{\beta}{(\beta + \bar{\pi}z\delta\bar{c})}c_{t+1} - \frac{\beta}{(R\beta + \bar{\pi}z\delta\bar{c})}\left(-\pi_{t+1} - z_{t+1}\right) \\ \pi_{t} &= \beta \frac{(\bar{\pi} - \pi_{*})}{2\bar{\pi} - \pi_{*}}\left[(c_{t+1} - c_{t}) + y_{t+1} - y_{t}\right] + \beta \pi_{t+1} + \left(\frac{\nu - (1 - \beta)\nu\phi\bar{\pi}(\bar{\pi} - \pi_{*})}{\nu\phi\bar{\pi}(2\bar{\pi} - \pi_{*})}\right)\nu_{t} \\ &+ \left(\frac{(1 - \nu)\bar{y}^{1 + 1/\eta}}{\nu\phi\bar{\pi}(2\bar{\pi} - \pi_{*})}\right)\left(c_{t} + \frac{1}{\eta}y_{t}\right) \\ c_{t} &= y_{t} - g_{t} \end{split}$$

Collecting terms and replacing the log-linearized resource constraint we have:

$$\hat{y}_{t} = \bar{\mathcal{D}}(\hat{y}_{t+1} - \hat{g}_{t+1}) + \bar{\mathcal{D}}(\hat{\pi}_{t+1} + \hat{z}_{t+1}) + \hat{g}_{t}$$

$$\hat{\pi}_{t} = \beta \hat{\pi}_{t+1} + \bar{\lambda}\nu_{t} + \bar{\kappa}(y_{t} - g_{t}) + \frac{\bar{\kappa}}{\eta}y_{t} + 2\bar{\Gamma}(y_{t+1} - y_{t}) - \bar{\Gamma}(g_{t+1} - g_{t})$$
(D.2)

Where $\bar{D} = \frac{\beta}{(\beta + \bar{\pi}z\delta\bar{c})}$, $\bar{\lambda} = \left(\frac{\nu - (1 - \beta)\nu\phi\bar{\pi}(\bar{\pi} - \pi_*)}{\nu\phi\bar{\pi}(2\bar{\pi} - \pi_*)}\right)$, $\bar{\kappa} = \left(\frac{(1 - \nu)\bar{y}^{1 + 1/\eta}}{\nu\phi\bar{\pi}(2\bar{\pi} - \pi_*)}\right)$, and $\bar{\Gamma} = \beta\frac{(\bar{\pi} - \pi_*)}{2\bar{\pi} - \pi_*}$, we obtain the log-linearized equations presented in the main text.

D.4. Proof of Proposition 5

Without shocks, the system of equations around a permanent liquidity trap can be rewritten as:

$$\hat{y}_{t} = \mathcal{D}\mathbb{E}_{t}(\hat{y}_{t+1} + \hat{\pi}_{t+1})$$
$$\hat{\pi}_{t} = \beta\mathbb{E}_{t}\hat{\pi}_{t+1} + \tilde{\kappa}\hat{y}_{t}$$
Where $\tilde{\kappa} = \left(\frac{1}{\eta} + 1\right) \left(\frac{(1-\nu)\bar{y}^{1+1/\eta}}{\nu\phi\bar{\pi}(2\bar{\pi}-\pi_{*})}\right), \bar{\mathcal{D}} = \frac{\beta}{\beta+\bar{\pi}z\delta\bar{c}}, \text{ then } 1/\bar{\mathcal{D}} = 1 + \frac{\bar{\pi}z\delta\bar{c}}{\beta}.$

To iterate the system forward, we write:

$$\begin{bmatrix} \bar{\mathcal{D}} & \bar{\mathcal{D}} \\ 0 & \beta \end{bmatrix} \begin{bmatrix} \hat{y}_{t+1} \\ \hat{\pi}_{t+1} \end{bmatrix} = \begin{bmatrix} 1 & 0 \\ -\tilde{\kappa} & 1 \end{bmatrix} \begin{bmatrix} \hat{y}_t \\ \hat{\pi}_t \end{bmatrix}$$

Then we can write:

$$\begin{bmatrix} \hat{y}_{t+1} \\ \hat{\pi}_{t+1} \end{bmatrix} = \begin{bmatrix} \frac{1}{\bar{\mathcal{D}}} & -\frac{1}{\bar{\beta}} \\ -0 & \frac{1}{\bar{\beta}} \end{bmatrix} \begin{bmatrix} 1 & 0 \\ -\tilde{\kappa} & 1 \end{bmatrix} \begin{bmatrix} \hat{y}_t \\ \hat{\pi}_t \end{bmatrix}$$

Simplifying:

$$\begin{bmatrix} \hat{y}_{t+1} \\ s\hat{\pi}_{t+1} \end{bmatrix} = \begin{bmatrix} \frac{1}{\bar{\mathcal{D}}} + \frac{\tilde{\kappa}}{\bar{\beta}} & -\frac{1}{\bar{\beta}} \\ -\frac{\tilde{\kappa}}{\bar{\beta}} & \frac{1}{\bar{\beta}} \end{bmatrix} \begin{bmatrix} \hat{y}_t \\ \hat{\pi}_t \end{bmatrix}$$

Define $\rho = 1/\beta, \phi = 1/\overline{D}$,

$$\begin{array}{c} \hat{y}_{t+1} \\ \hat{\pi}_{t+1} \end{array} \right] = \left[\begin{array}{c} (\phi + \tilde{\kappa}\rho) & -\rho \\ -\tilde{\kappa}\rho & \rho \end{array} \right] \left[\begin{array}{c} \hat{y}_t \\ \hat{\pi}_t \end{array} \right] + \left[\begin{array}{c} \frac{1}{\bar{\mathcal{D}}} & -\frac{1}{\bar{\beta}} \\ -0 & \frac{1}{\bar{\beta}} \end{array} \right] \left[\begin{array}{c} 0 \\ -\lambda \end{array} \right] \hat{v}_t$$

From this system, define $M \equiv \begin{bmatrix} (\phi + \tilde{\kappa}\rho) & -\rho \\ -\tilde{\kappa}\rho & \rho \end{bmatrix}$. Then, we can derive the following properties of the matrix *M*:

$$det(M) = \phi \rho, \quad tr(M) = \phi + (1 + \tilde{\kappa})\rho$$

Proposition C1 in (Woodford, 2003, pp 670) provides the necessary and sufficient conditions for determinacy for a system of 2 equations. A 2×2 matrix *M* with positive determinant has both eigenvalues outside the unit circle if and only if

$$\det M > 1, \quad \det M - \operatorname{tr} M > -1, \quad \det M + \operatorname{tr} M > -1$$

Under our sign restrictions and the assumption that $\beta < 1$, first and third inequalities necessarily hold, do that both eigenvalues are outside the unit circle if and only if $\phi > \frac{1-\rho(1+\tilde{\kappa})}{1-\rho} = 1 - \frac{\rho \tilde{\kappa}}{1-\rho}$ for determinacy. This implies $1/\bar{D} > \frac{\frac{\beta}{\beta} - \frac{1}{\beta}(1+\tilde{\kappa})}{\frac{\beta-1}{\beta}} = \frac{(\beta-1)-\tilde{\kappa}}{\beta-1} = 1 + \frac{\tilde{\kappa}}{1-\beta}$. We can rewrite this inequality to obtain $\frac{1-\beta}{1-\beta+\tilde{\kappa}} > \bar{D}$, which yields the restriction in the proposition.

D.5. MSV solution

This section derives the MSV solution for the analysis in Section 5.

D.5.1 proof of Proposition 6

We guess that $\hat{y}_t = a_1 \hat{z}_t + a_2 \hat{g}_t + a_3 \hat{v}_t$ and $\hat{\pi}_t = b_1 \hat{z}_t + b_2 \hat{g}_t + b_3 \hat{v}_t$ and solve for the unknown a's and b's. Replacing the guess into (D.2), collecting terms and using the method of undetermined coefficients, we obtain the following system of equations:

$$\begin{split} \hat{y}_{t} &= \bar{\mathcal{D}}(\mathbb{E}\hat{y}_{t+1} - g_{t+1} + E\hat{\pi}_{t+1} + \hat{E}z_{t+1}) + g_{t} \\ \hat{\pi}_{t} &= \beta \mathbb{E}\hat{\pi}_{t+1} + \bar{\kappa}(\frac{1}{\eta} + 1)\hat{y}_{t} - \bar{\kappa}\hat{g}_{t} + \bar{\lambda}\hat{v}_{t} + \bar{\varphi}(E\hat{g}_{t+1} - \hat{g}_{t}), \end{split}$$

where $\bar{\mathcal{D}} = \frac{\beta}{(\beta + \bar{\pi}z\delta\bar{c})}$, $\bar{\varphi} = \frac{(\bar{\pi} - \pi_*)}{2\bar{\pi} - \pi_*}$, $\bar{\lambda} = \left(\frac{\nu - (1 - \beta)\nu\phi\bar{\pi}(\bar{\pi} - \pi_*)}{\nu\phi\bar{\pi}(2\bar{\pi} - \pi_*)}\right)$, and $\bar{\kappa} = \left(\frac{(1 - \nu)\bar{y}^{1 + 1/\eta}}{\nu\phi\bar{\pi}(2\bar{\pi} - \pi_*)}\right)$. Guessing $y_t = a_1\hat{z}_t + a_2\hat{g}_t + a_3\hat{\nu}_t$ and $\pi_t = b_1\hat{z}_t + b_2\hat{g}_t + b_3\hat{\nu}_t$ we can replace back into the previous equations. Dropping time subscripts:

$$\begin{aligned} a_1 \hat{z}_t + a_2 \hat{g}_t + a_3 \hat{v}_t &= \bar{\mathcal{D}} \left(a_1 \rho_z \hat{z}_t + a_2 \rho_g \hat{g}_t + a_3 \rho_v \hat{v}_t - \rho_g g + b_1 \rho_z \hat{z}_t + b_2 \rho_g \hat{g}_t + b_3 \rho_v \hat{v}_t + \rho_z \hat{z}_t \right) + \hat{g}_t \\ &= \bar{\mathcal{D}} (a_1 \rho_z + b_1 \rho_z + \rho_z) \hat{z}_t + (\bar{\mathcal{D}} a_2 \rho_g - \bar{\mathcal{D}} \rho_g + \bar{\mathcal{D}} b_2 \rho_g + 1) \hat{g}_t + \bar{\mathcal{D}} \left(b_3 \rho_v + a_3 \rho_v \right) \hat{v}_t \end{aligned}$$

$$\begin{split} b_{1}\hat{z}_{t} + b_{2}\hat{g}_{t} + b_{3}\hat{v}_{t} &= \beta \left(b_{1}\rho_{z}\hat{z}_{t} + b_{2}\rho_{g}\hat{g}_{t} + b_{3}\rho_{v}\hat{v}_{t}\right) + \bar{\varphi}\left(\rho_{g} - 1\right)\hat{g}_{t} + \bar{\lambda}\hat{v}_{t} \\ &+ \bar{\kappa}(\frac{1}{\eta} + 1)\left(a_{1}\hat{z}_{t} + a_{2}\hat{g}_{t} + a_{3}\hat{v}_{t}\right) - \bar{\kappa}\hat{g}_{t} \\ &= \left(\beta b_{1}\rho_{z} + \bar{\kappa}(\frac{1}{\eta} + 1)a_{1}\right)\hat{z}_{t} + \left(\beta b_{2}\rho_{g} + \bar{\varphi}\left(\rho_{g} - 1\right) + \bar{\kappa}(\frac{1}{\eta} + 1)a_{2} - \bar{\kappa}\right)\hat{g}_{t} \\ &+ \left(\beta b_{3}\rho_{v} + \bar{\kappa}(\frac{1}{\eta} + 1)a_{3} + \bar{\lambda}\right)\hat{v}_{t} \end{split}$$

Comparing terms we can write the following system of equations:

$$\begin{bmatrix} 0\\ -\bar{\varphi}(\rho_g-1)+\bar{\kappa}\\ -\bar{\lambda}\\ -\bar{D}\rho_z\\ \bar{D}\rho_g-1\\ 0 \end{bmatrix} = \begin{bmatrix} (\beta\rho_z-1) & 0 & 0 & \bar{\kappa}(\frac{1}{\eta}+1) & 0 & 0\\ 0 & (\beta\rho_g-1) & 0 & 0 & \bar{\kappa}(\frac{1}{\eta}+1) & 0\\ 0 & 0 & (\beta\rho_v-1) & 0 & 0 & \bar{\kappa}(\frac{1}{\eta}+1)\\ \bar{D}\rho_z & 0 & 0 & (\bar{D}\rho_z-1) & 0 & 0\\ 0 & \bar{D}\rho_g & 0 & 0 & (\bar{D}\rho_g-1) & 0\\ 0 & 0 & \bar{D}\rho_v & 0 & 0 & (\bar{D}\rho_v-1) \end{bmatrix} \begin{bmatrix} b_1\\ b_2\\ b_3\\ a_1\\ a_2\\ a_3 \end{bmatrix}$$

,

The solution is:

$$b_1 = \frac{-\bar{\mathcal{D}}\rho_Z \bar{\kappa} \left(\frac{1}{\eta} + 1\right)}{\bar{\mathcal{D}}\rho_z \left[\bar{\kappa} \left(\frac{1}{\eta} + 1\right) + (1 - \beta\rho_z)\right] - (1 - \beta\rho_z)}; \qquad a_1 = \frac{(1 - \beta\rho_z)b_1}{\bar{\kappa} \left(\frac{1}{\eta} + 1\right)}$$
(D.3)

$$b_{2} = \frac{(1 - \bar{\mathcal{D}}\rho_{g}) \left[\bar{\varphi}(1 - \rho_{g}) - \bar{\kappa}\frac{1}{\eta}\right]}{\bar{\mathcal{D}}\rho_{g} \left[\bar{\kappa}\left(\frac{1}{\eta} + 1\right)\right] - (1 - \beta\rho_{g})(1 - \bar{\mathcal{D}}\rho_{g})}; \qquad a_{2} = \frac{\bar{\kappa} + (1 - \beta\rho_{g})b_{2} + \bar{\varphi}(1 - \rho_{g})}{\bar{\kappa}\left(\frac{1}{\eta} + 1\right)}$$
(D.4)

$$b_{3} = \frac{-\bar{\lambda}(1-\bar{\mathcal{D}})\rho_{v}}{\bar{\mathcal{D}}\rho_{v}\left[\bar{\kappa}\left(\frac{1}{\eta}+1\right)+(1-\beta\rho_{v})\right]-(1-\beta\rho_{v})}; \qquad a_{3} = \frac{\bar{\mathcal{D}}\rho_{v}b_{3}}{1-\bar{\mathcal{D}}\rho_{v}}$$
(D.5)

D.5.2 Unconditional Correlation: proof of Proposition 7

From the solution $\{a_i, b_i\} \forall i \in [1, 2, 3]$ derived in Proposition 6, we can see that inflation and output are positively correlated conditional on technology growth shocks and price-markup shocks. Positive correlation between inflation and output also obtains under government spending shocks if $\bar{\kappa} < \frac{\bar{\pi} z \bar{c} \delta(1-\beta)}{\bar{R} \beta}$.

Proof. We can use the matrix equations to alternately rewrite output and inflation IRF to govt spending shock as follows.

$$a_2 = \frac{1 + \mathcal{D}\rho_g(b_2 - 1)}{1 - \bar{\mathcal{D}}\rho_g}$$

Consequently, $a_2 > 0$ whenever $b_2 > 0$.

When $b_2 < 0$, a condition that guarantees that $a_2 < 0$ is $b_2 < -\frac{\bar{\kappa}}{(1-\beta)}$ (From D.4 and the fact that $\bar{\varphi} < 0$). Rewrite this condition, and substitute in the values of parameters to obtain the requirement that $\bar{\kappa} < \frac{\bar{\pi}z\bar{c}\delta(1-\beta)}{\bar{R}\beta}$ is sufficient for positive correlation between inflation and output.

D.5.3 Posterior distribution of model weights





Notes: The solid black line is the posterior mean of λ estimated recursively over the period 1998:Q1-2020:Q1. The shaded areas correspond to the 90 percent credible set of the posterior distribution.

D.6. Prior distributions

Table 4 lists the priors used for estimation of the DSGE model of Section 4, including information on the marginal prior distributions for the estimated parameters. Under the prior, we assume that all estimated parameters are distributed independently which implies that the joint prior distribution can be computed from the product of the marginal distributions.

Parameters	Description	Distribution	P(1)	P(2)
ρ_g	Persistence gov. spending shock	${\mathcal B}$	0.6	0.2
ρ_{ν}	Persistence markup shock	${\mathcal B}$	0.6	0.2
ρ_z	Persistence technology. growth shock	${\mathcal B}$	0.4	0.1
σ_{g}	Std dev. gov. spending shock	\mathcal{IG}	0.004	Inf
σ_{ν}	Std dev. markup shock	\mathcal{IG}	0.004	Inf
σ_{z}	Std dev. markup shock	\mathcal{IG}	0.004	Inf
σ_{ζ}	Std dev. sunspot shock	\mathcal{IG}	0.004	Inf
$corr(\epsilon_z, \epsilon_{\zeta})$	Correl. sunspot	\mathcal{U}	0	0.5774
$corr(\epsilon_{\nu}, \epsilon_{\zeta})$	Correl. sunspot	\mathcal{U}	0	0.5774
$corr(\epsilon_g, \epsilon_{\zeta})$	Correl. sunspot	\mathcal{U}	0	0.5774

Table 4: Prior Distribution of DSGE parameters

Notes: G is Gamma distribution; B is Beta distribution; IG is Inverse Gamma distribution; and U is Uniform distribution. P(1) and P(2) are mean and standard deviations for Beta, Gamma, and Uniform distributions.

D.7. Posterior sampler

We can solve the log-linearized system of equations of Section 4 using standard perturbation techniques. As a result, the likelihood function can be evaluated with the Kalman filter. We generate draws from the posterior distribution using the random walk Metropolis algorithm (RWM) described in An and Schorfheide (2007). We scale the covariance matrix of the proposal distribution in the RWM algorithm to obtain an acceptance rate of approximately 60%. For posterior inference we generated 50,000 draws from the posterior distribution and discard the first 25,000 draws.

D.8. Comparative Statics: Expectations Trap vs Secular Stagnation

The BSGU and the secular stagnation hypotheses have contrasting implications for shocks and policy. These differences stem from the local determinacy property of these steady states, which translate into differences in slopes of aggregate supply and aggregate demand in our model. We now demonstrate these properties with comparative static experiments. Because of local determinacy of the secular stagnation steady state, the comparative static experiment is well-defined without the need for additional assumptions. With the BSGU steady state, we assume that inflation expectations do not change drastically to push the economy to the full-employment steady state in response to the experiment.

In Figure 9, solid lines plot the steady-state AD-AS representation of the quantitative model under two parametrizations. Annualized inflation deviation relative to the central



Figure 9: Permanent increase in markups v

bank target inflation is on the vertical axes and output gap relative to target-steady state output (in percents) is on the horizontal axes. The coordinates (y^b, π^b) and (y^s, π^s) denote the expectations-driven and fundamentals-driven liquidity trap steady states respectively. The left panel plots AD-AS curves when prices are relatively flexible (κ^b) and the natural rate of interest is positive. The AD-AS intersection depicted at (y^b, π^b) is locally indeterminate, features zero nominal interest rates and output is permanently below potential. At this intersection, the AS curve is steeper than the AD curve. In the right panel, we plot the AS curve with relatively rigid prices (κ^s) , and the AD curve with negative natural interest rate, $r^* < 0$. AD intersects AS at the secular stagnation steady state at the coordinate (y^s, π^s) .

An upward shift in aggregate supply curve in Figure 9, denoted with dashed blue line, induced by permanent increase in steady state markups, translates into higher output under secular stagnation and lower output under BSGU. Under secular stagnation, the natural interest rate is too low for the central bank to stabilize the economy. An increase in markups through inflationary pressures helps lower real interest rate, thus reducing the real interest rate gap and expand output. Under BSGU, the problem is of pessimism about inflation expectations. If agents remain pessimistic about inflation undershooting its target, an increase in markups is further contractionary since the resource inefficiencies associated with increased markups dominate the increase in output demand due to higher prices (see also Mertens and Ravn, 2014).

An outward shift in aggregate demand in Figure 10, denoted with dashed red line, induced by permanent increase in steady state TFP growth, translates to higher output under secular stagnation but lower output under BSGU. Higher TFP growth signals higher income for households and leads to increased consumption demand. This increased impatience translates into higher output under secular stagnation. Under BSGU, in contrast, the increased TFP



Figure 10: *Permanent increase in TFP growth rate z*

growth translates into higher reduction in prices by firms, which dominates the increased demand by households. As a result, there is lower output and inflation under BSGU.

Similarly, a neo-Fisherian exit policy of raising interest rates at the ZLB is contractionary under secular stagnation as it increases the real interest rate gap from natural rate, but it is expansionary at the BSGU steady state equilibrium (Schmitt-Grohé and Uribe, 2017). Furthermore, an increase in government expenditure (financed by lumpsum taxes) or a permanent reduction in short term interest rates below the ZLB has inflationary effects under secular stagnation but deflationary effects under BSGU.²⁹

These disparate policy implications raise the question whether it is possible to distinguish these two different kinds of liquidity traps in the data. We turn to this question next.

²⁹We model the neo-Fisherian policy as a permanent change in the intercept of the Taylor rule, *a*: $R^{new} = \max\{1 + a, a + R^* \left(\frac{\Pi}{\Pi^*}\right)^{\phi_{\pi}}\} = a + R$. where *a* is increased to a positive number from zero. This policy simultaneously increases the lower bound on nominal interest rate and thus does not have any effect on the placement of the kink in the aggregate demand curve. Given the inflation rate, an increase in *a* lowers output demanded. At the secular stagnation steady state, this induces deflationary pressures that increases the real interest rate gap and causes a further drop in output. In contrast, during a BSGU trap, an increase in nominal interest rate anchors agents' expectations to higher levels of inflation, thus obtaining the neo-Fisherian results (Schmitt-Grohé and Uribe, 2017). The effects of increased government spending on output are somewhat ambiguous because of elastic labor supply that also causes changes in the aggregate supply curve.

E. MEDIUM-SCALE DSGE MODEL

We present the equilibrium conditions of the model written in the form of stationary variables. Let A_t be the non-stationary level of TFP at time t. We normalize the following variables :

$$y_t = Y_t / A_t,$$

$$c_t = C_t / A_t,$$

$$k_t = K_t / A_t,$$

$$k_t^u = K_t^u / A_{t-1},$$

$$\mathbb{I}_t = I_t / A_t,$$

$$w_t = W_t / (A_t P_t),$$

$$r_t^k = R_t^k / P_t,$$

$$\lambda_t = \Lambda_t A_t,$$

Definition 1 (Normalized equilibrium). 17 endogenous variables $\{\lambda_t, i_t, c_t, y_t, \Pi_t, mc_t, \tilde{\Pi}_{t-1}, \Pi_t^w, \tilde{\Pi}_{t-1}^w, w_t, L_t, k_{t+1}^u, r_t^K, \mathbb{I}_t, q_t, u_t, k_t\}$, 6 endogenous shock processes $\{z_t, g_t, \eta_t^b, \mu_t, \nu_{p,t}, \nu_{w,t}\}$, 6 exogenous shocks $\{\epsilon_{z,t}, \epsilon_{g,t}, \epsilon_{\eta^b,t}, \epsilon_{\mu,t}, \epsilon_{\nu_{w,t}}, \epsilon_{v_{w,t}}\}$ given initial values of k_{t-1}^u .

Consumption Euler equation

$$\lambda_t = \beta(1+i_t) \mathbb{E}_t \left[\frac{\lambda_{t+1}}{z_{t+1}} \frac{1}{\Pi_{t+1}} \right] + \delta_t , \qquad (E.1)$$

$$\lambda_{t} = \frac{1}{c_{t} - \frac{hc_{t-1}}{z_{t}}} - h\beta \mathbb{E}_{t} \frac{1}{z_{t+1} \left[c_{t+1} - \frac{hc_{t}}{z_{t+1}} \right]},$$
(E.2)

Price-setting

$$(1 - \nu_{p,t}) - mc_t + \nu_{p,t}\psi_p \left(\frac{\Pi_t}{\tilde{\Pi}_{t-1}} - 1\right)\frac{\Pi_t}{\tilde{\Pi}_{t-1}} - \nu_{p,t}\psi_p\beta E_t\frac{\lambda_{t+1}}{\lambda_t}\left(\frac{\Pi_{t+1}}{\tilde{\Pi}_t} - 1\right)\frac{\Pi_{t+1}}{\tilde{\Pi}_t}\frac{y_{t+1}}{y_t} = 0$$
(E.3)

$$\tilde{\Pi}_{t-1} = \bar{\Pi}^{1-\iota_p} \Pi_{t-1}^{\iota_p}$$
(E.4)

Wage-setting

$$\nu_{w,t}\psi_{w}\left[\frac{\Pi_{t}^{w}}{\tilde{\Pi}_{t-1}^{w}}-1\right]\frac{\Pi_{t}^{w}}{\tilde{\Pi}_{t-1}^{w}}=\nu_{w,t}\psi_{w}\beta\mathbb{E}_{t}\left[\frac{\Pi_{t+1}^{w}}{\tilde{\Pi}_{t}^{w}}-1\right]\frac{\Pi_{t+1}^{w}}{\tilde{\Pi}_{t}^{w}}+L_{t}\lambda_{t}\left[\omega\frac{L_{t}^{\frac{1}{\eta}}}{\lambda_{t}}-(1-\nu_{w,t})w_{t}\right]$$
(E.5)

$$\tilde{\Pi}_{t-1}^{w} = z \bar{\Pi}^{1-\iota_{w}} \left(\exp(\epsilon_{z,t}) \Pi_{t-1} \right)^{\iota_{w}}$$
(E.6)

$$\Pi_{W,t} = \frac{w_t}{w_{t-1}} \Pi_t z_t , \qquad (E.7)$$

Capital investment

$$k_{t+1}^{u} = \mu_t \left[1 - S\left(\frac{\mathbb{I}_t}{\mathbb{I}_{t-1}} \frac{z_t}{z}\right) \right] \mathbb{I}_t + (1 - \delta_k) \frac{k_t^u}{z_t},$$
(E.8)

$$q_{t} = \beta \mathbb{E}_{t} \left[\frac{\lambda_{t+1}}{\lambda_{t} z_{t+1}} \left(r_{t+1}^{K} u_{t+1} - a(u_{t+1}) + q_{t+1}(1 - \delta_{k}) \right) \right],$$
(E.9)

$$q_{t}\mu_{t}\left[1-S\left(\frac{\mathbb{I}_{t}}{\mathbb{I}_{t-1}}\frac{z_{t}}{z}\right)-S'\left(\frac{\mathbb{I}_{t}}{\mathbb{I}_{t-1}}\frac{z_{t}}{z}\right)\frac{\mathbb{I}_{t}}{\mathbb{I}_{t-1}}\frac{z_{t}}{z}\right]$$
$$+\beta\mathbb{E}_{t}\left[\mu_{t+1}\frac{\lambda_{t+1}}{\lambda_{t}}q_{t+1}\frac{z_{t+1}}{z}\left(\frac{\mathbb{I}_{t+1}}{\mathbb{I}_{t}}\right)^{2}S'\left(\frac{\mathbb{I}_{t+1}}{\mathbb{I}_{t}}\frac{z_{t+1}}{z}\right)\right]=1$$
(E.10)

Capital utilization rate

$$k_t = u_t \frac{k_t^u}{z_t},\tag{E.11}$$

$$r_t^K = a'(u_t)$$
, (E.12)

Production technologies

$$y_t = A_t k_t^{\alpha} L_t^{1-\alpha} , \qquad (E.13)$$

$$r_t^k = \alpha m c_t \frac{y_t}{k_t}, \qquad (E.14)$$

$$w_t = (1 - \alpha)mc_t \frac{y_t}{L_t}, \qquad (E.15)$$

Government

$$\frac{1+i_t}{1+i_{ss}} = \max\left(\frac{1}{1+i_{ss}}, \left(\frac{1+i_{t-1}}{1+i_{ss}}\right)^{\rho_R} \left[\left(\frac{\Pi_t}{\bar{\Pi}}\right)^{\phi_\pi} \left(\frac{y_t z_t}{z y_{t-1}}\right)^{\phi_{dy}}\right]^{1-\rho_R} \exp(\epsilon_{mp,t})\right), \quad (E.16)$$

Market clearing

$$y_t = c_t + \mathbb{I}_t + a(u_t)\frac{k_t^u}{z_t} + \left(1 - \frac{1}{g_t}\right)y_t$$
, (E.17)

Law of motion of Shocks The six structural shocks driving the model economy are assumed to follow first order auto-regressive processes of the form $\log(x_t) = (1 - \rho_x) \log(x) + \rho_x \log(x_{t-1}) + \sigma_x \epsilon_{x,t}$, with $\epsilon_{x,t} \sim N(0, 1)$, and x denoting steady-state values, for $x = z, g, \eta^b, \mu, \nu_p, \nu_w$.

Parameters Table 5 details the values of the parameters for the medium-scale DSGE model of section 6.

Parameter	Description	Value
β	Discount factor	0.942
δ	Marginal utility of bonds	0.109
ν_p	Price markup parameter	0.22
ν_w	Wage markup parameter	0.2
α	Capital elasticity of output	0.37
δ_k	Capital Depreciation rate	0.015
h	Consumption habit	0.358
ν	Inverse of Frisch labor elasticity	2.27
ω	Labor disutility	0.492
σ_a	Capital utilization elasticity	2.246
ψ_I	Investment adjustment cost	5.16
ψ_p	Price adjustment cost	1600
ψ_w	Wage adjustment cost	833.73
ι_p	Indexation on price inflation	-0.225
ι_w	Indexation on wage inflation	-0.295
$ ho_w$	Persistence wage-markup shock	0.252
$ ho_p$	Persistence price-markup shock	0.268
ρ_g	Persistence government spending shock	0.843
$ ho_{\mu}$	Persistence investment shock	0.353
ρ_z	Persistence technology shock	0
$ ho_b$	Persistence risk premium shock	0.488
$100\sigma_w$	Standard deviation wage-markup shock	0.338
$100\sigma_p$	Standard deviation price-markup shock	0.394
$100\sigma_g$	Standard deviation government spending shock	0.45
$100\sigma_{\mu}$	Standard deviation investment shock	4.269
$100\sigma_z$	Standard deviation technology shock	1.641
$100\sigma_b$	Standard deviation risk premium shock	0.351

 Table 5: Parameters Medium-Scale DSGE Model

F. SIMPLE MODEL WITH DOWNWARD NOMINAL WAGE RIGIDTY

We now derive the results presented in Section 2 with wage-setting frictions and inelastic labor instead of price-rigidities and elastic labor. Our model is a variant of the downward-nominal

wage rigidity apparatus introduced by Schmitt-Grohé and Uribe (2017).

Time is discrete and there is no uncertainty. Suppose the representative agent supplies labor $\bar{h} = 1$ inelastically and maximizes the following utility function choosing consumption good C_t and one-period (real) risk-free government bonds b_t :

$$\max_{\{C_t, b_t\}} \mathbb{E}_0 \sum_{t=0}^{\infty} \theta_t \left[\log C_t \right]$$
$$\theta_0 = 1;$$
$$\theta_{t+1} = \hat{\beta}(\tilde{C}_t) \theta_t \forall t \ge 0$$

where θ_t is an endogenous discount factor (Uzawa, 1968; Epstein and Hynes, 1983), C_t is consumption, \tilde{C}_t is average consumption that the household takes as given, and h_t is hours. For tractability, we assume a linear functional form for $\hat{\beta}(\cdot) = \delta_t \beta C_t$, where $0 < \beta < 1$ is a parameter, and $\delta_t > 0$ are exogenous shocks to the discount factor. The household earns wage income $W_t h_t$, interest income on past bond holdings of risk-free government bonds b_{t-1} at gross nominal interest rate R_{t-1} , dividends Φ_t from firms' ownership and makes transfers T_t to the government. Π_t denotes gross inflation rate. The period by period (real) budget constraint faced by the household is given by

$$C_t + b_t = \frac{W_t}{P_t} h_t + \frac{R_{t-1}}{\Pi_t} b_{t-1} + \Phi_t + T_t$$

An interior solution to household optimization yields the Euler equation:

$$1 = \beta(\tilde{C}_t) \mathbb{E}_t \left[\frac{C_t}{C_{t+1}} \frac{R_t}{\Pi_{t+1}} \right]$$

In equilibrium, individual and average per consumption are identical, i.e. $C_t = \tilde{C}_t$. The Euler equation simplifies to:

$$1 = \delta_t \beta C_t \mathbb{E}_t \left[\frac{C_t}{C_{t+1}} \frac{R_t}{\Pi_{t+1}} \right]$$

Consumption goods are produced by competitive firms with labor as the only input using the technology

$$Y_t = F(h_t) = h_t^{\alpha}$$
, where $0 < \alpha < 1$

These firms set price of the final good P_t to equate marginal product of labor to the marginal cost.

$$F'(h_t) = \frac{W_t}{P_t}$$

We introduce a very stylized form of downward nominal wage rigidity (following Schmitt-Grohé and Uribe 2017):

$$W_t \ge (1 - \kappa + \kappa (1 - u_t)^{\alpha}) W_{t-1} \equiv \tilde{\gamma}(u_t) W_{t-1}$$

where $\kappa > 0$, and $u_t \equiv 1 - \frac{h_t}{h}$ is involuntary unemployment. This downward rigidity implies that employment cannot exceed the total labor supply in the economy i.e. $h_t \leq 1$. We further assume that the following slackness condition holds:

$$(\bar{h} - h_t)(W_t - \tilde{\gamma}(u_t)W_{t-1}) = 0$$

We close the model by assuming a government that balances budget,

$$b_t + T_t = \frac{R_{t-1}}{\Pi_t} b_{t-1}$$

and a monetary authority that sets nominal interest rate on the net zero supply of nominal risk-free one-period bonds using the following Taylor rule

$$R_t = \max\{1, (1+r^*)\Pi_t^{\varphi_\pi}\}$$

where $(1 + r^*) \equiv \frac{1}{\delta_t \beta}$ is the natural interest rate, and $\phi_{\pi} > 1.3^{\circ}$ The zero lower bound (ZLB) constraint on the short-term nominal interest rate introduces an additional nonlinearity in the policy rule. Finally, we assume that the resource constraints hold in the aggregate:

$$C_t = Y_t$$
, and $b_t = 0$.

F.1. Equilibrium

Let $w_t \equiv \frac{W_t}{P_t}$ denote the real wage. The competitive equilibrium is given by the sequence of seven endogenous processes $\{C_t, Y_t, R_t, \Pi_t, h_t, w_t, u_t\}$ that satisfy the conditions (F.1) - (F.7) for a given exogenous sequence of process $\{\delta_t\}_{t=0}^{\infty}$ and the initial condition w_{-1} :

$$1 = \delta_t \beta C_t \mathbb{E}_t \left[\frac{C_t}{C_{t+1}} \frac{R_t}{\Pi_{t+1}} \right]$$
(F.1)

$$Y_t = h_t^{\alpha}.$$
 (F.2)

$$\alpha h_t^{\alpha - 1} = w_t \tag{F.3}$$

$$h_t \le \bar{h}, \quad w_t \ge (1 - \kappa + \kappa h_t^{\alpha}) \frac{w_{t-1}}{\Pi_t}, \quad (\bar{h} - h_t) \left(w_t - (1 - \kappa + \kappa h_t^{\alpha}) \frac{w_{t-1}}{\Pi_t} \right) = 0$$
 (F.4)

$$u_t = 1 - \frac{h_t}{\bar{h}} \tag{F.5}$$

$$R_t = \max\{1, (1+r^*)\Pi_t^{\phi_\pi}\}\}$$
(F.6)

$$Y_t = C_t \tag{F.7}$$

where the exogenous sequence of natural interest rate is given by $1 + r_t^* \equiv \frac{1}{\delta_t \beta}$.

³⁰The natural interest rate is defined as the real interest rate on one-period government bonds that would prevail in the absence of nominal rigidities.

F.2. Non-stochastic steady state

In the steady state, we can simplify the system of equations to an aggregate demand block and an aggregate supply block.

Aggregate Demand (AD) is a relation between output and inflation and is derived by combining the Euler equation and the Taylor rule. Mathematically, the AD curve is given by

$$Y_{AD} = \frac{1}{\beta \delta} \begin{cases} \frac{1}{(1+r^*)\Pi^{\phi \pi - 1}}, & \text{if } R > 1, \\ \Pi, & \text{if } R = 1 \end{cases}$$
(F.8)

When ZLB is not binding, the AD curve has a strictly negative slope, and it becomes linear and upward sloping when the nominal interest rate is constrained by the ZLB. The kink in the aggregate demand curve occurs at the inflation rate at which monetary policy is constrained by the ZLB: $\Pi_{kink} = \left(\frac{1}{(1+r^*)}\right)^{\frac{1}{\phi\pi}}$. When $1 + r^* > 1$, the kink in the AD curve occurs at an inflation rate below 1. For the natural interest rate to be positive, the patience parameter must be low enough i.e. $\delta < \frac{1}{\beta}$. The dashed red line in panel a) of Figure 11 plots the aggregate demand curve with a positive natural rate.

Aggregate Supply (AS): Because of the assumptions of downward nominal wage rigidity and capacity constraints on production, the AS curve features a kink at full employment level of output and gross inflation rate equal to one. When inflation rate is less than one, the downward wage rigidity constraint becomes binding. As a result, inflation cannot fall to completely adjust any demand deficiency and firms layoff workers. The aggregate supply curve can be summarized by:

$$Y_{AS} \le 1, \quad \Pi \ge (1-\kappa) + \kappa Y_{AS}, \quad (Y_{AS} - 1) \left(1 - (1 - \kappa + \kappa Y_{AS}) \frac{1}{\Pi} \right) = 0$$
 (F.9)

When $h = \bar{h} \equiv 1$, $\Pi \ge 1$. The AS curve is a vertical line at full employment. For h < 1, $\Pi = (1 - \kappa + \kappa h^{\alpha})$. The AS curve is linear and upward sloping with slope= κ for y < 1. The kink in the AS curve occurs at the coordinate Y = 1, $\Pi = 1$. Because of this assumed linear aggregate supply curve under deflation, the degree of nominal rigidity κ also determines the lower bound on inflation $(1 - \kappa)$. The solid blue line in both panels of Figure 11 plots the aggregate supply curve.

Note that the equilibrium conditions are similar to those presented in Section 2, with the major exception being that the steady state AS graph is vertical at h = 1 due to the upper bound on labor endowment. One can derive similar results as in Section 2. We only note the following proposition to complete proofs for statements regarding robust minimum wage policy in Section 2.8.

Proposition 8. (Minimum wage policy): Let $0 < \delta < \frac{1}{\beta}$. A minimum income policy that installs a lower bound $1 - \kappa$ on nominal wage growth can preclude the expectations trap.





Proof. The downward sloping portion of AD curve goes through Y = 1, $\Pi = 1$, and so does the vertical portion of the AS curve. When $0 < \delta < \frac{1}{\beta}$, $1 + r^* > 1$. The kink in the AD curve occurs at inflation rate below 1. Thus, there always exists an intersection between the AS and the AD at Y = 1 and $\Pi = 1$. To show that there does not exist another equilibrium, note that, AD is linear and upward sloping when ZLB is binding and AS is also linear and upward sloping for gross inflation below 1. Furthermore for $\Pi_{kink} \leq \Pi < 1$, $Y_{AD} > 1 > Y_{AS}$. And for $\Pi = 1 - \kappa$, $Y_{AS} = 0 < Y_{AD}$. Thus there does not exist a steady state with zero nominal interest rate.

G. EXPECTATIONS TRAP IN AN OLG MODEL

The degree of nominal rigidities also plays a key role in eliminating the locally indeterminate stagnation steady state in the overlapping generations model of Eggertsson et al. (2019) (EMR). We outline the key message here while referring the reader to EMR for a detailed model. Agents live for three periods: young, middle, and old. Young are borrowing constrained and derive no income. Middle supply labor inelastically to perfectly competitive firms for wages and save for retirement. Old consume the savings made when middle. Supply and demand for savings results in the following bond market-clearing condition : $1 + r_t = \frac{1+\beta}{\beta} \frac{D_t}{Y_t - D_{t-1}}$, where *D* is the exogenous debt limit faced by the young borrowers. It is further assumed that households do not accept nominal wages below a particular wage norm i.e. $W_t = \max{\{\bar{W}_t, W_t^{flex}\}}$ where $\bar{W}_t = \gamma W_{t-1} + (1 - \gamma)W_t^{flex}$ and $W_t^{flex} = P_t \alpha$. Perfectly competitive firms hire workers to produce final output using production function $Y_t = h_t^{\alpha}$, taking wages as given. The policy rule is the same as in our baseline model in Section 2.

Given inflation target $\Pi^* = 1$, the aggregate demand and the aggregate supply blocks in the steady state are given by:

$$Y_{AD} = D + \begin{cases} \frac{1+\beta}{\beta} \frac{D}{\Gamma^*} \Pi^{1-\alpha_{\pi}}, & \text{if } \mathbb{R} > 1, \\ \frac{1+\beta}{\beta} D\Pi, & \text{if } \mathbb{R} = 1 \end{cases}$$
(G.1)

where $\Gamma^* = (1 + r^*)^{-1}$.

$$Y_{AS} = \begin{cases} 1, & \text{if } \Pi \ge 1, \\ \left(\frac{1-\frac{\gamma}{\Pi}}{1-\gamma}\right)^{\frac{\alpha}{1-\alpha}}, & \text{if } \Pi < 1 \end{cases}$$
(G.2)

If $\Pi^* = 1$, $r^* > 0$, and $\gamma < 0$, then there exists a unique liquidity trap steady state with positive unemployment, deflation, and zero nominal interest rate. The dynamics around this steady state are locally indeterminate. A negative value of γ implies that nominal wages increasingly fall with unemployment as in Schmitt-Grohé and Uribe (2017).³¹

H. COMPARISON WITH THE TEXTBOOK EULER EQUATION

We provide a brief comparison of results for the reader with the textbook Euler equation (Woodford, 2003). We illustrate the role of two central elements in our framework - a) the modified Euler equation and b) bounds on deflation.

In the textbook model, the natural interest rate is always fixed at $\frac{1}{\beta} > 1$. As a result, the aggregate demand relationship is a horizontal line at $\Pi = \beta < 1$ when the ZLB constrains the nominal interest rate. However, the existence of an *unintended* deflationary steady state is contingent on the assumptions regarding the supply side of the economy. Suppose the y-intercept of the aggregate supply curve is large enough. In that case, there does not exist a deflationary steady state.³² Setting this y-intercept is analogous to a minimum wage policy discussed in Section 2.8.

While modifying the Euler equation does not eliminate the expectations-trap steady state, it is necessary for the secular stagnation steady state to exist. The modified Euler equation with an endogenous long-run natural interest rate opens up the possibility of a secular stagnation steady state. This steady state cannot arise in the standard model because of a violation of the transversality condition of the representative household.

Bonds-in-utility is an alternative way to introduce discounting in the Euler equation (Michaillat and Saez, 2021; Michau, 2018; Ono and Yamada, 2018). Time variation in the preference for bonds, captured by δ in our model, has a functional equivalence with risk-premium in medium-scale DSGE models (Fisher, 2015).³³ Another interpretation of the shocks to δ is

³¹Ascari and Bonchi (2020) study the use of income or wage growth policies to reflate an economy experiencing persistent ZLB in the EMR model.

³²In the notation of Schmitt-Grohé and Uribe 2017, if $\gamma(1) > \tilde{\beta}$ there does not exist an unemployment steady state in their baseline model.

³³It is straightforward to show similar functional equivalence between preferences with endogenous

that these capture the *flight to liquidity* episode of the recent financial crisis (Krishnamurthy and Vissing-Jorgensen, 2012). A similar wedge in the Euler equation can be associated with the deterioration in liquidity properties of AAA-rated corporate bonds in contrast to Treasury securities during the 2008-09 financial crisis (Del Negro et al., 2017).

While these different interpretations follow naturally from extensive work in the secular stagnation literature. We view this wedge in the Euler equation as a reduced-form representation of microfoundations such as population aging, savings glut, reserve accumulation, inequality, or debt deleveraging (Eggertsson et al. 2016, 2019; Auclert and Rognlie 2018). Remaining agnostic about the origin of this wedge allows researchers to easily introduce the secular stagnation hypothesis in DSGE models.³⁴

I. DECREASING MARGINAL IMPATIENCE WITH CAPITAL ACCUMULATION

For completeness, we sketch the setup to show existence of a stable equilibrium with Uzawa-Epstein preferences under decreasing marginal impatience. We switch to continuous time formulation to analytically show stability

I.1. Simple model of Secular Stagnation

First, we abstract from capital and shut down any asset accumulation. Household maximizes the following lifetime utility

$$\int_0^\infty e^{-\int_0^t \rho(c_v)dv} u(c_t)dt$$

where $\int_0^t \rho(c_v) dv$ is the time-preference term that depends on the past and present consumption function through the function ρ . The function $\rho(c_t)$ denotes the instantaneous rate of time preference and it depends on c_t alone.

Assumption 1. (Das 2003) The functions u(c) and $\rho(c)$ are real valued bounded above and twice continuously differentiable on $(0, \infty)$, and

1. for all
$$c > 0$$
, $u(c) > 0$; $u'(c) > 0$; $u''(c) < 0$

2. for all c > 0, $\rho(c) > 0$, $\rho'(c) < 0$; and $\rho''(c) \le 0$. Also $\rho(0) = \overline{\rho}$; the upper bound on ρ .

These assumptions are standard, except for decreasing marginal impatience $\rho'(c) < 0$. Note that the upper bound on ρ will be crucial to guarantee the existence of a unique equilibrium.

The budget constraint of the household is:

$$\dot{a}_t = r_t a_t + w_t - c_t$$

discounting, and preferences with wealth in the utility function.

³⁴In the recent literature that augments DSGE models with endogenous growth, (mean zero) shocks to preference for bonds are added to get co-movement of investment and consumption as well to derive the *divine coincidence* benchmark (Garga and Singh, 2021).

where a(0) = 0. Let $\theta(t) = \int_0^t \rho(c_v) dv$. Then $\dot{\theta} = \rho(c_t)$ and $\theta(0) = 0$. Thus the optimization exercise of the household is

$$\max \int_0^\infty u(c_t) e^{-\theta(t)} dt$$

subject to the budget constraint $\dot{a}_t = r_t a_t + w_t - c_t$, $\dot{\theta} = \rho(c_t)$, a(0) = 0 and $\theta(0) = 0$.

The Hamiltonian function is

$$H = u(c_t)e^{-\theta(t)} + \gamma \left(r_t a_t + w_t - c_t\right) - \lambda(\rho(c))$$

where γ and $-\lambda$ are the co-state variables associated with the two variables *a* and θ , respectively. Rescaling the co-state variables so that the terms involving θ get eliminated, we can write the first-order conditions as

$$u'(c) - \mu - \phi \rho'(c) = 0$$
 (I.1)

$$\dot{\mu} = \mu \left[\rho(c) - r_t \right] \tag{I.2}$$

$$\dot{a}_t = r_t a_t + w_t - c_t \tag{I.3}$$

$$\dot{\phi} = \phi \rho(c) - u(c) \tag{I.4}$$

where $\mu = \gamma e^{\theta}$ and $\phi = \lambda e^{\theta}$ are the new co-state variables. Differentiating the first condition wrt time and rewriting:

$$\dot{c} = \frac{u'(c) - \phi \rho'(c)}{-u''(c) + \phi \rho''(c)} \left[r - \rho(c) - \rho'(c) \frac{\phi \rho(c) - u(c)}{u'(c) - \phi \rho'(c)} \right]$$
(I.5)

Assuming $\rho(C) = \bar{\rho} - \frac{\kappa}{u'(C)}$ and $u(c) = \log(c)$. Thus $\rho''(c) = 0$, $\rho(0) = \bar{\rho}$, and $\rho'(c) = \kappa$. thus, Equations I.3, I.4 and I.5, represent a system of differential equations involving three variables *a*, *c*, and ϕ . In the steady state. $\dot{c} = 0$, $\dot{a} = 0$, and $\dot{\phi} = 0$. The steady state is thus characterized by the following equations:

$$\phi = \frac{u(c)}{\rho(c)} \tag{I.6}$$

$$c = ra + w, \tag{I.7}$$

$$r - \rho(c) - \rho'(c) \frac{\phi \rho(c) - u(c)}{u'(c) - \phi \rho'(c)} = 0$$
(I.8)

These simplify to give us:

$$r = \rho(c) = \bar{\rho} - \kappa C$$

This is exactly same as equation 4 in Michau (2018). A strictly positive increase in κ implies a finite elasticity of steady state consumption *c* with respect to steady state interest rate *r*. Note that we shut down wealth accumulation a = 0 in our model to generate this steady state. As in Michau, we also assume zero pure profits. Presence of pure profits will slightly modify the household budget constraint without any loss of results shown here. Following the arguments in Section 2 of Michau (2018), one can show the existence of secular stagnation steady state.

I.2. A model of Secular Stagnation with capital

Household wealth is composed of physical capital K_t , and government bonds B_t .

$$\dot{a}_t = r_t a_t + w_t l_t + \psi_t - c_t$$

where ψ_t are real lump-sum transfers. household wealth will also include capital K_t , with real price $p_t^K = \frac{P_t^K}{P_t}$. Households are indifferent between holding risk-free capital and risk-free government bonds. Hence no-arbitrage implies that the two assets must yield the same return

$$r_t = \frac{R_t/P_t}{p_t^K} - \delta + \frac{\dot{p}_t^K}{p_t^K}$$

where δ is the capital depreciation rate. The intertemporal budget constraint prevents households from running Ponzi schemes:

$$\lim_{t\to\infty}e^{-\int_0^t r_s ds}a_t\geq 0$$

We will have the same utility function for the representative household as in the above subsection:

$$\max \int_0^\infty u(c_t) e^{-\theta(t)} dt$$

The first-order conditions by the the maximum principle are:

$$\dot{c} = \frac{u'(c) - \phi \rho'(c)}{-u''(c) + \phi \rho''(c)} \left[r - \rho(c) - \rho'(c) \frac{\phi \rho(c) - u(c)}{u'(c) - \phi \rho'(c)} \right]$$
(I.9)

$$\dot{\phi} = \phi \rho(c) - u(c) \tag{I.10}$$

$$\dot{a}_t = r_t a_t + w_t l_t + \psi_t - c_t \tag{I.11}$$

along with the transversality condition

$$\max \int_0^\infty u(c_t) e^{-\theta(t)} dt$$

Steady state equilibria
$$\phi = \frac{u(c)}{\rho(c)} \tag{I.12}$$

$$c = F(L, K) - \delta K, \tag{I.13}$$

$$r - \rho(c) - \rho'(c) \frac{\phi \rho(c) - u(c)}{u'(c) - \phi \rho'(c)} = 0$$
(I.14)

$$r = i - \pi \tag{I.15}$$

$$r_t = F_K - \delta \tag{I.16}$$

$$w_t = \frac{W_t}{P_t} = \begin{cases} F_L, & \text{if } \Pi \ge 1, \\ \frac{\kappa}{\Pi + \kappa - 1}, & \text{if } \Pi \in [1 - \kappa, 1] \end{cases}$$
(I.17)

Let f(k) = F(K, L)/L, where production function *F* is assumed to be homogenous in labor. As in Das (2003), we assume that

$$-f''(k) > -\rho'(f(k) - \delta k)(f'(k) - \delta).$$

This condition states that the marginal return to capital decreases faster than marginal impatience. As in Schmitt-Grohé and Uribe (2003), if we further assume that the discounting is exogenous to individual agents, the Euler equation simplifies to

$$\dot{c} = \frac{u'(c)}{-u''(c)}[r - \rho(c)]$$

The resulting system is observationally equivalent to that of Michau (2018) which shows existence of secular stagnation steady state.